MIDDLE CLASS DETERMINANTS IN LATIN AMERICA (2000-2010)

A GENDER PERSPECTIVE

Emmanuelle Martinez, and Gwenn Parent* * Paris School of Economics - French Ministry of Labor ¹ March 1st, 2012 DRAFT PAPER

Abstract

This paper investigates the determinants of middle class using multinomial probit modeling the likelihood of being poor (living below 50% of national median income), middle class (between 50% and 150%) or affluent (above 150%) and related gender specific effects for six Latin American countries (Argentina, Bolivia, Chile, Colombia, Mexico and Peru).

The analysis finds that middle class determinants in Latin America do not differ significantly from poverty ones, although with weaker effects. Female heads of households mostly belong to the poor and middle class. All countries show a similar pattern, with the older heading wealthier households. As expected, education remains a powerful, although with weakening over time, determinant of income classes belonging: primary education determines poor and middle class status while secondary and tertiary education increases the likelihood of being affluent.

The only noticeable difference between poverty and middle class determinants concerns employment characteristics (activity status): heads of household being unemployed, inactive or working in agriculture or public and social services leads household to be poor (but not middle class nor of course affluent). On the contrary, other household's members being unemployed or inactive leads households to be poor or middle class. Self-employment is another important determinant of poverty but not of middle class.

Adopting a gender perspective using two alternative specifications of multinomial probit, the analysis finds that (i) female headed households more often belong to the poor and middle sector (ii) education benefits them more than their male counterparts (iii) and they suffer less from being unemployed or inactive (in a majority of the sample countries). Husbands' characteristics have a greater impact on household's income class belonging than their spouses' characteristics. Spouses being inactive or self-employed are determinants of poverty and middle class while their counterpart husbands are only determinants of poverty. Somehow strikingly inactive as well as active spouses are negatively impacted by their husbands not working (inactive or unemployed). Indeed in terms of husbands characteristics' effect on spouses, no noticeable differences are found between countries where spouses are mostly active (such as in Bolivia and Peru) and countries where they are mostly inactive (such as in Chile).

JEL Classification: J16, I3, D3, D6

Keywords: Middle class determinants, gender, social status, income distribution, Latin America.

¹ This document reflects exclusively the views of the authors and not of the Paris School of Economics, or the French Ministry of Labor. Contact information: Gwenn Parent <u>gwenn.parent@pse.ens.fr</u> (corresponding author) and Emmanuelle Martinez <u>emmanuellemartinezs@hotmail.com</u>. The authors would like to thank the IADB for its interest in the subject, and the support it provided to the redaction of this article. We are very grateful to Francesca Castellani and Jannet Zenteno.

I. Introduction

The literature shows a growing interest in the concept of middle class over the past decade, especially in Latin American countries, where middle class size increased significantly. While the notion of belonging to the middle class appears to be universally attractive, it is not immediately clear what being "middle-class" means. In particular, do the characteristics of the middle class transcend national borders? This paper tackles some of these issues, applying an income-based definition of the middle class, to analyse the characteristics that determine class belonging across Latin American countries.

Most authors question the middle class definition and more specifically the relevance of having an absolute definition (taking into account different thresholds) or a relative one. They also deal with the significance of different policies –fiscal, social, educational- to be undertaken to prevent the middle class from falling into poverty. Here, this paper aims to underline what characteristics affect the likelihood of households' social classes belonging (belong to the middle sector rather than being poor or affluent). Indeed the underlying idea is to put into relief the weaknesses and strengths of the middle class so as to know what put them into the danger of impoverishment and on the contrary what strengthen them, so as to design the appropriate policies. We then focus on the gender aspects of social classes belonging.

The literature focuses on poverty determinants with no comparison with the other social classes (e.g. middle sector and affluent). In this paper, we use multinomial probit models for six Latin American countries (e.g. Argentina, Bolivia, Chile, Colombia, Mexico and Peru) to characterize the main determinants that can be considered as middle class strengths and weaknesses, so as to suggest public policies in favor of middle sector's sustainability.

The organization of the paper goes as follows, section II discusses middle class measurement, section III describes the main results outlined by the literature. Then section IV presents our methodology and empirical data used. Our main results are presented in section V, first regarding multinomial probit on heads of households, then adopting a gender perspective focusing on female specific determinants, and finally considering models based only on spouses to determine the relative importance of husband's and spouses' characteristics in income classes belonging. Section VI concludes.

II. How to define middle class

As mentioned, we used a median income based definition. Broadly speaking, measures to estimate middle class size can be categorized as to whether they rely on economic and/or social criteria. The first refers to an income/consumption range that segments population distribution². The second group includes reference to specific behavioral characteristics, as education, occupational status, and consumption patterns. Opinion surveys constitute an alternative way to identify middle classes.³

Income based definitions are usually either "absolute or relative". The former assumes fixed (i.e. absolute) income ranges (PPP adjusted); the latter middle ranges of national income distributions (i.e. quintiles). While income-based definitions enjoy higher analytical rigor than perception-based concepts, they remain debatable. Absolute thresholds suffer from some arbitrariness,⁴ which emerges when applied to heterogeneous development levels. An absolute benchmark, while providing a common reference, might fall short of accounting for country-specific features and development levels. Relative definitions might provide less homogeneous boundaries as they are country tailored. In general, absolute definitions have been applied to the evolution of the global middle class while relative boundaries for country specific investigation.

To gauge middle class size "measurement-sensitivity", Castellani and Parent (2011) compare several alternative measurements and apply them to selected OECD and Latin American countries: ⁵

- a) PPP based definition 2- 20 USD (2005 PPP) per capita per day
- b) Distribution based definition: leaving out the poorest 20% and the richest 20%

c) Median income based definition: 50-150% of median income (i.e. poverty generally defined as 50-60% of median income)

d) Poverty-line based definition: lower bound the national poverty line (national, urban), NPL and the upper bound is set as a multiple (3 times) the poverty line

Given that distribution based definition results in a fixed middle class size (60%) and that poverty line definition might imply different national methodologies, measurements based on median income seem to provide some degree of homogeneity, allowing for cross country comparison and reflecting within-country distribution patterns. Moreover, being invariant to changes in income, it can be easily used to analyze size evolution over time.

Besides income, the debate over the middle class is often focused on standards of living. This entails ownership of durable goods, educational levels, and occupational status. A middle class life style is typically associated with the ownership of equipment goods (i.e. refrigerators, telephones, automobiles), to such an extent that several contributions to the middle class literature consider the durable goods ownership as a 'defining gauge' for the stage of development of a country. A look at these elements helps identify "features" that might define a global middle class beyond income levels. Literature on social determinants has mostly dealt with the determinants of poverty; however, its methodology remains similar to that of studying middle class. The following section presents the literature results.

² Easterly (2001), Thurow (1999), Byrdsall (2001), Eisenhauer (2008)

³ Latinobarómetro

⁴ While poverty thresholds are clearly defined, middle class "boundaries" rely on arbitrary limits

⁵ See also Byrdsall et all (2000).

III. Literature review

The middle class concept and measurement as well as its particularities in comparison with the other social classes (poor and affluent) are still widely discussed and contended. It is of interest to underpin recent ways of measuring the middle class (distinguishable from the ones mentioned in the previous section), although as it was the case with previous measures, all authors agree on the fact that the size of the middle class in Latin America has increased over time, and still is.

The polarization-based measurement, that relies on the fact that individuals identify with a group and feel alienation with respect to other groups depending on their incomes, exhibits homogenous results in terms of middle class evolution across the years (Cruces et al., 2011).

They find that through this measurement the middle class in Latin America⁶ has increased over time, that their level of education is in between that of the poor and the affluent and that the middle class is closer to the affluent in terms of both employment (that is higher among the middle class and the affluent) and unemployment (that is higher among the poor). The recent middle class measures also include the notion of vulnerability of the social class. Torche and Lopez-Calva (2010) define the middle class according to household characteristics related to the capacity to generate income and show that the middle class remains characterized by some degree of economic vulnerability which reduces the differences between them and the poor in spite of different income levels. In the same vein, Lopez-Calva and Ortiz-Juarez (2011) exhibit a view of the middle class in Latin America⁷ has increased during the past two decades. According to them, a quite large amount of people above the poverty line should not be classified as middle class since their vulnerability to poverty is high, and despite the fact that part of the population do not fall under the classification of middle class given their vulnerability to poverty, they are not eligible for poverty programs according to traditional definitions while they would need it.

All these middle class measures based on income have also lead to some conclusions regarding the homogeneity of the middle class regarding values on society. Lopez-Calva and al. (2011) show that there is a robust link between the income-based middle class and their values, although the middle class does not have many specific values linked to belonging to this social class. Indeed, except from its moderation the middle class values are in between that of the poor and the affluent, and varies more according to the countries than to the social class.

Other authors have questioned the different middle class income-based definitions. Cardenas, Kharas and Heano (2011) argue that attempts to use income measures are limited by the lack of information about incomes in values surveys which is either absent or classified in broad categories. Some authors focus on the subjective measure of the middle class. Using the Gallup World Survey, Lora and Fajardo (2010) make a comparison between income-based middle class definitions and self-perceived status. Their results show that median income based definition displays the higher mismatched with self-perceived status while absolute income displays the best matching. The differences between income-based measurements and self-perceived status are that the latest is associated not only with income but also with personal capabilities, interpersonal

⁶ The sample used by the authors includes Argentina, Brazil, Chile, El Salvador, Mexico and Uruguay.

⁷ The sample used by the authors includes Chile, Mexico and Peru.

relations, financial assets and the perception of economic insecurity. In the same way, Amoranto, Chun and Deolalikar (2010) also rely on the self-perception of either status or position in the income distribution to measure the middle class in emerging countries.

Poverty measurement has also been questioned. Bourguignon and Chakravarty (2003) opened a new way in the area with the introduction of the concept of multidimensional poverty. Indeed, they exhibit that poverty should not only be defined by income or consumption because it is a multidimensional concept. They reject the previous literature attempts that tried to aggregate various attributes into a single index and a unique poverty line. Indeed they propose a specific poverty line for each dimension of poverty and consider that one is poor if he or she falls below at least one of this line. Alkine and Santos (2011) followed the way and constructed a poverty index using household level data aggregated at the country level. Just as the rest of the literature, Alkine and Foster (2007) take virtually all the elements of the measures as given (determined outside measure), notably the dimension of poverty, the dimension specific cutoffs, the weighs on deprivation and the minimum number of deprivation needed to be deemed "poor". Poverty measurement literature has gone in this direction since then but Ravallion (2011) argues that there still remain flaws in the measurement. Poverty monitoring efforts should rely on a credible set of indices each considering a different dimension of poverty relevant to a specific setting rather than one. He questions the notions of weighs attributed to each poverty attribute that should not be selected by the economist but consistent with well-informed choices made by people.

While the concepts of middle class and poverty measurement are widely discussed, the literature mostly deals with the determinants of poverty and extreme poverty and very little with the determinants of the middle class.

Most of the literature is mainly based on the study of poverty determinants in various developing countries, showing differences in terms of methodology but similar choice of variables. Most authors take into account household composition and individual characteristics and some of them also include housing and/or access, to community services or specific countries determinants. Household composition commonly includes the number of children by age class, the number of elderly, and whether the household is headed by a female. Individual characteristics include age, gender, education (the decomposition of the variable varies given the papers), employment (despite some differences authors commonly take into account employment, unemployment, inactivity and self-employment) and sectors of activity. Common results point that education, being employed, being male, and age positively affect income and decrease poverty.

Some authors include housing characteristics (the tenure of housing and its characteristics). Housing ownership can be very closely linked to income, although Fiess and Verner (2004) argue that in Brazil, in 1998, housing ownership in urban areas among the extreme poor was 78% against 70% for the affluent. Properties of the household are interesting as they represent the household's inventory of wealth and therefore affect its income flow. Furthermore, some households, especially in rural areas, can be poor in terms of income, but wealthy when their property is taken into consideration.

Fiess and Verner (2004) study poverty in Northeastern Brazil in comparison with the rest of the country based on a probit analysis. They take into account the following housing characteristics: tenure, materials of roof and walls, access to electricity, safe water and sanitation.

Using a sample of Latin American countries⁸ from 1986 to 1998 and country-specific household survey data, Wodon et al. (2010) perform an analysis of the impact of poverty determinants on the log of per capita income divided by the poverty line. Apart from the determinants mentioned above, they include the variable ethnicity that displays an important negative impact, and they also consider migration that shows a positive effect. Their study of individual determinants indicates that the spouse's characteristics often have a smaller impact than the ones of the head of household on poverty.

As they mention, households headed by females display a negative impact on poverty and spouses' characteristics are less important than their husbands'. Therefore, females' decisions-making is of great importance in the analysis of their belonging to various social classes.

The analytical framework of females' decisions that will impact household income includes female decisions of participating in the labor market, getting married and with whom; having children and how many years of education they receive.

Mizala et al. (1999) in a Chilean study based on household surveys show that women labor participation positively depends on being head of household, having daughters between 19 and 24 years old living at home, having a maid and beneficiating from house facilities (mostly running water and electricity). However women labor participation is negatively correlated with being married, having children under 18 years old living at home independently of their gender. Connely and al. (1996) had previously found similar results on Brazil, as well as Wong and Levin (1992) on Mexico. More recently, Atal (2010) using a general equilibrium model taking into account decision making intra household, having children, beneficiating from tax credit and technological improvements validates the same findings, and also confirms Meyer and Rosenbaum (2001) positive impact of income tax credit on women labor participation, as well as Schonberg et al. (2007) negative impact of maternity leave on women getting back to labor market.

The decision of having children and getting married seems to be linked both to education and income. Through an IADB network dynamic analysis of household decision making in Latin America encompassing different country studies (Brazil, Colombia, Mexico, Peru and Uruguay) ran by different authors, Attanasio et al. (2002) show that having children is decreasing with the level of education while it is positively linked to income.

The level of education in Latin America tends to be higher than it used years ago and negatively impacts marriage; indeed educated women get married latter if sometimes not at all. Blau et al. (2000), through a USA study, put in evidence that white better female labor markets (16-24 years old) and black better female labor market (25-34 years old) tend to lower marriage rates.

⁸ Mexico, Guatemala, El Salvador, Hondura, Nicaragua, Colombia, Bolivia, Chile and Brazil.

All of these results show that the income brought to the household by females depends on female's decision-making but it is also closely linked to the existence of gender gaps such as earning gender gaps and educational gender gaps that are the most commonly analyzed in the literature.

The increase of educational level in Latin America and the reverse educational gender gap most countries of the region have experienced, surely have an interesting impact on female belonging to middle class. Indeed, Duryea et al. (2007) exhibit that in Latin America the schooling gender gap has closed for the cohort born at the end of the 60's and reversed for the following cohorts such that females have on average a quarter of schooling more than males. The changes are explained by educational attainment of females at the higher levels rather than improvements in the early years of education. However Bolivia, Guatemala, Mexico and Peru have not yet closed the gap in adult schooling attainment and the noticeable gender differences regard indigenous boys and girls of the lowest income quintiles. One of the plausible implications of these changes in the gender educational gap has to do with changes in marriage markets. It has been documented that individuals are delaying marriage decisions across the world, including in Latin America. Furthermore there has been an expanding education in Latin America for the past two decades and the proportion of the labor force with at least secondary education has increased from 40% to 60% (Gasparini et al., 2011, Galvis, 2010), females beneficiating from this expansion at least as much as males.

Another element that probably accounts for part of females' head of household specific characteristics is gender earning gap. The vast literature on the subject has underlined that disparities are partially attributed to gender differences in observable socio-demographic (age, education, occupation, working time, experience, type of contracts, job status, marital status, having children), job characteristics (level of formality, occupational segregation by gender) and cultural, political norms or geographic location. It is has been exhibited that earning gender gap is not homogeneously distributed but affects women at the top of the income distribution (Kandil, 2009) or at the top and the bottom of the income distribution (Badel et al. 2009,). When in relation with education, the top of the distribution regards tertiary educated females and the bottom low educated ones (De la Rica, 2005). Regarding the socio-demographic differences, females decisions mentioned previously play a predominant role. However, after comparing males and females with the same characteristics, Nopo et al. (2011) puts into relief that earning wage gap falls between a range of 8% to 48% but remains consistent. Therefore an important component of those earning differentials cannot be explained by gender differences in observable characteristics, is not linked to females' decision-making and remains to be explored. Unexplained pay gap is higher among older, informal and self-employed workers (Atal et al. 2009). In Latin America, males still earn 9-27% more than females with high cross country heterogeneity (Atal et al. 2009). Indeed, the global gender gap index that benchmarks national gender gaps on economy, politic, education and healthbased criteria underlines the heterogeneity between the countries (Hausmann et al. 2010).9

Female heads of household often display a negative impact on their household's income in the literature, while the case of spouses is little analyzed. Given the importance of both females 'decision-making and gender gaps they suffer from on their individual and household characteristics (marital status, education, type of contract, number of children, status...), after

⁹ Positions of the Latin American countries of our sample among the 135 positions (1 being the country with the smaller gender gaps and 135 the country with the worst) Argentina: 28, Chile: 46; Bolivia: 62, Peru: 73, Colombia: 80 and Mexico: 89

having examined the determinants of the head of households' income classes belonging in the first place, we decide to focus on a comparison of males and females head of household in a second one so as to analyze the specific characteristics of female heads of household and the impact of these characteristics on their income classes belonging. Finally and in the third place, given the fact that females head (as explained in the following sections) only represent a few percentage of all females, we also focus on spouses to identify the impact of their characteristics on the household income classes belonging in comparison with the ones of their husband. And we try to analyze whether spouses play a role in the determination of the household's income classes belonging.

IV. Methodology and data presentation

For a binary treatment analysis such as poverty belonging, logit and probit models usually leads to similar results. However, when leaving the binary treatment case, the choice of the model becomes more important. The multiple treatment analysis is constituted of more than two alternatives (see Imbens (2000) and Lechner (2001)). The multinomial probit (MNP) model plays an important role in social, econometric and biological sciences for the analysis of multi-category response. It enables a greater degree of flexibility in modelling discrete categories over the commonly adopted multinomial logit (MNL) model, based on stronger assumptions than the MNP, making the latter one the preferable option¹⁰. The main advantage of MNP is that it allows an analysis of multiple, unordered outcomes. Nevertheless, our variable class is obviously ordered, but unlike ordered models, the MNP estimate allows different coefficients for every dependant variable on each poverty/middle class/affluent outcome¹¹. Despite the loss of the easily-computed form of the MNL model, we preferred the multinomial probit model in the present analysis.

The variable of interest, "income class" is based on the total household income adjusted for family composition with OECD adult equivalent scale¹², allowing income comparison of households of different sizes and structures. The analysis considers three possible outcomes for each household: 1) poor households, e.g. with income¹³ lower than 50% of national median income, 2) middle class households, e.g. earning between 50% and 150% of median income, and 3) affluent households with more than 150% of median income.

¹⁰ Especially the `independence from irrelevant alternatives' assumption (IIA) is critical. It basically states that the odds ratio between two alternatives are independent of other alternatives. This assumption is convenient for estimation as it is less burdensome to compute but not appealing from an economic or behavioural point of view (for details see e.g. Greene (2003)).

¹¹ We computed as well multinomial logit and ordered probit on Colombia (2010), and found relatively stable results no matter the selected model. All marginal effects have the same sign, with relatively stable size.

¹² This is the "OECD-modified scale", which has been adopted by the European Commission, among others. Other scales used in international comparisons include the square root of household size (used in many OECD studies since the 1990s). In practice, the difference implied by the choice of one or another of these weighting schemes is small. See Castellani and Parent (2010) for more details.

¹³ Income adjusted for family composition

Our discussion of the determinants of middle class builds on the use of household surveys for Argentina, Bolivia, Chile, Mexico and Peru. Colombian results are computed on Living Standards Measurement Study surveys (LSMS). ¹⁴

In the early 2000 Chile and Mexico had the smallest poor class (20%) of all 6 countries in our panel, while Colombia had the biggest one (37% of households were poor in Colombia in 2003). Poverty has declined in all six countries over the past decade but not homogenously: in 2009-2010, Chile and Mexico still have the smallest poor class (around 17%); Colombia experienced a huge reduction of poverty (-17pp in 7 years); Bolivia is the new bad student of LAC in 2009/2010 with 23% of poverty.

In the early 2000's, Colombia had the smallest middle class (24% in 2003), while Mexico and Chile had the biggest one (48% in 2000). Ten years later, the progression of middle class is impressive, especially for Colombia (+22pp in 7 years): in 2009/2010, middle class households represent from 45% in Bolivia up to 53% in Mexico, three other countries counting with more than half of total households being middle class: Argentina, Chile and Peru.

Finally, in the early 2000's, Peru had the smallest affluent class (31%), while Colombia had the biggest one (39% in 2003). The proportion of affluent households declined down to 30 to 33% of total households in 2009-2010. Affluent class size is therefore relatively homogenous across countries at the end of the period.

We estimate three different specifications of multinomial probit models (MNP):

The first specification, taking into consideration all head of households, investigates the impact of the head of household's individual characteristics , as well as his employment, occupation and economic ones on income classes belonging. In a second specification, we focus on gender specific effects for female headed households, authorising specific gender effects for all individual and work-related characteristics. Finally, our last specification is based on a different population: we run MNP on head of household's spouses including both their own individual and work characteristics and their husband's characteristics. The aim of this last model is to identify the relative importance of spouses' decisions and characteristics compared to the head of household ones.

Model 1 can be specified as follows:

$$Class_{i} = \alpha_{1}Gender_{i} + \beta_{1}X_{i} + \gamma_{1}H_{i} + \varepsilon_{1i}$$
(1)

where $Class_i$ is the income class of household *i* (either poor, middle class or affluent), *Gender_i* a dummy variable for household head gender, X_i a set of individual characteristics of the head of household *i*, H_i a set of employment, occupation and economic variables of the head of household. We assume ε_{1i} to be *i.i.d* $N(0, \sigma_1^2)$.

¹⁴ Annex 1 lists the countries, and the survey data's source. It also lists the sample sizes and the proportions of poor, middle class and affluent households in each survey.

The set of dependent variables we chose as possible determinants of poverty, middle class, and affluent belonging may be describe as follows:

- The vector of individual characteristics X_i includes age classes, educational attainment (primary education, secondary or technical and university), marital status (single or couple including non-married couples) and a dummy variable for ethnic origin (majority or ethnic minority group).
- The vector of employment, occupation and other economic characteristics H_i includes a dummy variable for geographic localization (living in the metropolitan region or not), the number of others household members active occupied, unemployed and inactive, a variable including activity status and sector of activity (when occupied) of the head of household and a dummy variable of self-employment of household head.

Our reference population considers households headed by a man, 41-65 years old, with primary education (completed or not), single, belonging to the majority ethnic group, who is occupied in services, not self-employed and who lives in the metropolitan region (Region of Gran Buenos Aires in Argentina, department of La Paz in Bolivia, metropolitan region of Santiago in Chile, district of Bogotá in Colombia, federal district and state of México in Mexico and region of Lima and Callao in Peru).

Other characteristics would have been interesting to include in this model, but for two types of reasons we couldn't consider them in our estimation:

Some variables would have been interesting determinants of class belonging, but were not available in our surveys: 1) Regional characteristics on the one hand, such as isolation/remoteness, weather (e.g.: are typhoons or droughts common) and environmental conditions (e.g. frequency of earthquakes), regional governance and management, and 2) Community characteristics, such as infrastructure (e.g. is there piped water, access to a tarred road), land distribution, access to public goods and services (e.g. proximity of schools, clinics), social structure and social capital, on the other, matter and may be of importance to determine the middle class determinants. ¹⁵.

We also eliminate variables with possible endogeneity problems regarding income such as house ownership, assets (land, tenure...), resource base including land and quality, health... Some variables are also endogenous to the "adjusted" income we used in the construction of classes: households characteristics are indeed already included in the construction of income adjusted for family composition (household size, composition of the household, age of each member)¹⁶.

¹⁵ Other important information such as safety, peace of mind, sustainable environment, belonging to a community could not be taken into account but might be relevant characteristics of the middle class ¹⁶ The only characteristics related to household structure we included are activity status of household members, which are not endogenous to the adjusted income.

Beyond the analysis of middle class determinants, this paper focuses on a gender perspective through two additional empirical models:

<u>Model 2</u> focuses on gender specific effects for female headed households, authorising specific gender effects for all individual and employment characteristics. Model 2 can be specified as follows:

$$Class_{i} = \alpha_{2}Gender_{i} + \beta_{2}Gender_{i} * X_{i} + \gamma_{2}Gender_{i} * H_{i} + \varepsilon_{2i}$$
(2)

where $Class_i$ is the income class of household *i* (either poor, middle class or affluent), *Gender_i* the household head gender, X_i the same set of individual characteristics as in model (1), H_i the same set of employment, occupation and economic variables as in model (1). We assume ε_{2i} to be *i.i.d* $N(0, \sigma_2^2)$. Considering *Gender_i* * X_i and *Gender_i* * H_i enables gender specific coefficients β_2 and γ_2 for individual and employment characteristics of female heads of household. It is thus interesting to compare male coefficients and female ones in order to point out gender disparities existing in middle class belonging.

<u>Model 3</u> is based on a different population than models 1 and 2: it focuses on spouses. The aim of this last model is to identify the relative importance of spouses' decisions and characteristics compared to their husband's ones. The empirical framework is a multinomial probit estimation on spouses only including both their own individual and work characteristics and their husband's characteristics. Model 3 that can be specified as follows:

$$Class_i = \beta_3 X_i + \gamma_3 H_i + \lambda_3 K_i + \varepsilon_{3i}$$
(3)

where $Class_i$ is the income class of household *i* (either poor, middle class or affluent), X_i the same set of individual characteristics as in model (1) for spouses, H_i the same set of employment, occupation and economic variables as in model (1) for spouses, and K_i a vector of individual and employment characteristics for the husband in household *i* (including husband's educational attainment, activity status and sector of activity, and self-employment). We assume ε_{3i} to be *i.i.d* $N(0, \sigma_3^{-2})$. Considering K_i enables comparison between the relative importance of spouses' own individual and employment characteristics and the ones of their husband on the determination of the household standard of living.

Next section presents and discusses our main findings.

V. Results

1. <u>Multinomial probit of the likelihood of being poor, middle sector or affluent</u>

Annex 2 presents all marginal effects of model 1 multinomial probits for six Latin American countries: Bolivia and Chile for 2009, Argentina, Colombia, Mexico and Peru for 2010. The following table synthesises our main findings:

				2009 / 2010
VARIABLES	Poor	Middle class	Affluent	Exceptions / notes
Male	Ref.	Ref.	Ref.	Bolivia, gender effect on poverty not significant (due to the very low
Female	+	+	-	proportion of households headed by women in Bolivia)
up to 30 years old	+	+	-	
31-40 years old	+	+	-	Argenting, elderly are more represented in middle class (+ on MC)
41-65 years old	Ref.	Ref.	Ref.	Argentina, eldeny are more represented in middle class (+ on MC)
more than 65 years old	-	-	+	
Primary education	Ref.	Ref.	Ref.	
Secondary education	-	-	+	Observed in all 6 countries
Technical education or University	-	-	+	
Single	Ref.	Ref.	Ref.	Argentina, Chile, Colombia: being in couple reduces the likelihood of
Couple	+	+	-	being poor (- on poor)
Majority group	Ref.	Ref.	Ref.	Effect unclear on middle class, variable not available on Argonting
Ethnic group	+	n.s.	-	Effect unclear off midule class, variable for available off Argentina
Other provinces/regions	Ref.	Ref.	Ref.	Polivia, gondor offect on poverty not significant
Capital	-	-	+	Bolivia, gender enect on poverty not significant
Number of other occupied household members	-	n.s.	+	Observed in all 6 countries
Number of other unemployed household members	+	+	-	Observed in all 6 countries
Number of other inactive household members	+	+	-	Observed in all 6 countries
Active occupied, agriculture	+	-	-	Chile: head of household working in agriculture increases the likelihood of being middle class
Active occupied, industry	n.s.	n.s.	n.s.	Chile and Mexico : + on middle class and - on affluent
Active occupied, trade	Variat	ole across co	ountries	Chile and Mexico: + on middle class. Argentine and Chile: - on affluent, Argentien and Peru: + on poverty
Active occupied, public and social services	+	-	variable across countries	Bovlie, Colombie, Mexico: n.s. or - on poverty, Chile: + on MC
Active occupied, other services	Ref.	Ref.	Ref.	
Active unemployed	+	- 1	-	Observed in all 6 countries
Inactive	+	- 1	-	Observed in all 6 countries
No self employed	Ref.	Ref.	Ref.	Chile: all the contrary: - on poor and MC, + on affluent
Sellemployeu	+	<u> </u>	-	

Synthesis of marginal effects sign and significance in MNP model 1

<u>Table 1:</u> Synthesis of marginal effects signs and significance in MNP Model 1

Middle class determinants in Latin America are found to be very close to poverty determinants, although in lower proportions (table1). Female heads of households mostly belong to the poor and middle class. In all countries, the age pattern is similar, the older the heads of household, the more likely they become affluent. As expected, education remains a powerful, although declining over time, determinant of income classes, primary education remaining a poor and middle class determinant while secondary and tertiary education increasing the likelihood of being affluent.

The only noticeable exception concerns employment characteristics (activity status): being unemployed, inactive or working in agriculture leads to poverty but not to middle class. Self-employment is another important exclusive determinant of poverty, but not of middle class.

Model 1 has also been computed for the period 2000-2006 in order to analyze results consistency and time evolution of middle class determinants. Detailed results are presented below:

<u>Gender:</u> Female is a determinant of poverty and middle sector, while male heads of household are more represented in the affluent class.

Regardless of the country, female-headed households are noticeably more likely to belong to the poor or the middle class (annex 3A). Indeed in all six countries, gender displays a negative impact on income classes belonging. The strongest impact is identified in Colombia while the smallest is in Peru (significant only at the end of the period, and for the middle sector and affluent part of the analysis), although the differences between countries are not too consistent.

Gender is a determinant of both poverty and middle class in Latin America: compared to the affluent, women tend to be over-represented in poor and middle sectors classes. In 2009/2010, being female increases by 4.4% (Mexico) to 6.0% (Argentina) the likelihood of being poor, by 3.3% (Argentina) to 9.6% (Bolivia) the likelihood of belonging to the middle sector, and decreases by - 5.0% (Mexico) to -11.4% (Colombia) the likelihood of being affluent. Over time, all things being equals, the disadvantage of being a female tends to decrease in Colombia (regarding poverty essentially), while it increases in Argentina and Mexico. The effect has remained quite stable in Chile since 2000.

<u>Age:</u> Younger household heads are more likely to be poor and in a lesser extent belong to the middle sector. The older they are, the more likely they will become affluent.

In all countries, the age pattern is similar, the older the heads of household, the more likely to climb the income ladder (annex 3B). Age displays the same pattern with a negative impact of being young (less than 30 years old and between 31-40 years old) and a positive one of being old (more than 65 years old).

In 2009 or 2010, less than 30 years old heads of household are between 5 and 15% more likely to be poor and between 4 and 11% more likely to belong to the middle sectors compared to the reference population, 41-65 years old. Being young decreases the likelihood of being affluent by between 10 and 20%. Elder heads of household (more than 65 years old) are generally less likely to be poor and more likely to be affluent.

It is interesting to note that in Bolivia, older heads of household are not particularly wealthier and young heads of household are not significantly poorer than in other Latin American countries. In Mexico, being old was not associated with poverty in the early 2000's, but it is since 2006. Last but not least, elderly heads of household are more likely to belong to the middle sector in Argentina while the contrary is observed in all 5 other Latin American countries studied.

Young households generally lie at the lower end of the income distribution.

<u>Education</u>: Primary education is a determinant of poverty and middle sector, secondary and tertiary education tends to pull towards affluent class.

As expected, education determines income: primary education is an important determinant of poverty and middle class, while secondary and tertiary education increases the likelihood of being affluent (annex 3C). Obviously, tertiary education displays a greater impact than secondary education.

In 2009-2010, poor and middle sector heads of household are mostly primary educated in Latin American countries: for example 80% of Bolivian poor heads of household are primary educated, as are 54% of Bolivian middle classers.

The highest impacts of education (both secondary and tertiary) on income classes belonging are observed in Chile, Colombia and Mexico. .

All other things being equal, we observe a decreasing impact of education along the years for almost all countries, which can be attributed to the fact a wider access to education. Education becoming more important for higher income classes in Argentina and Colombia since 2003 might be linked to the recovery those countries after the end of the 90s crisis, although this hypothesis cannot be validated here.

<u>Matrimonial status</u>: Marital status effect is ambiguous on poor and affluent sectors but increases the likelihood of belonging to the middle sector.

Interpreting matrimonial status is quite challenging because the effect captured in MNP includes various effects on income classes, due to the way these classes were constructed (cf. methodology section):

- Being in couple may bring a second salary to the household's income composition but household composition affects adjusted income "per capita". Couples have a greater probability of having children, so the construction of income adjusted for family composition will mechanically lower household income by dividing total earning by the sum of total household members weights (cf. methodology section). For example, for a household with two children (one younger than 14, one older) and two parents we consider total earnings divided by 2.3. This might have a negative impact on income classes belonging for couples with children.
- Endogamy must also play a role here: as individuals tends to marry someone close to their own origin, social category, education... couple marginal effects may capture individual characteristics effects of the partner, reinforcing the disparities between classes: indeed affluent heads of household in couple are more likely to be with a better educated partner, of the same origin. In this case endogamy will induce to capture all "positive" effects of such good characteristics in the "couple" effect. On the other hand, poor heads of household, if married, are more likely to live with a less educated partner. Couple effect in this case will penalize him more, capturing these "penalizing" individual characteristics of the partner.

In conclusion, final effect of the couple determinant is therefore ambiguous (annex 3D), depending on the relative importance of each of these mechanisms.

In Argentina and Chile in 2009-2010, being in couple decreases the likelihood of being poor by -2.9% (Chile) to -4.1% (Argentina), increases the likelihood of being middle class by 2.4% (Argentina, not significant in Chile) and that of being affluent by 1.7% (Argentina) to 1.8% (Chile). On the contrary, being in couple increases the likelihood of being poor in Bolivia, Mexico, Peru and Colombia. It is interesting to note that whatever the country (the impact being positive or negative regarding the poor and the affluent) being in couple always increases the likelihood of belonging to the middle sector.

The variable is not consistent all years in all five countries although it is tricky to draw conclusions regarding the tendency along the years.

<u>Origin:</u> Minority ethnicity is a determinant of poverty, while belonging to the majority group is a determinant of affluent class.

Belonging to an ethnic minority group displays a negative impact in all 6 countries of the sample (annex 3E). In 2009-2010, it increases the likelihood of being poor, while minority households are less likely to be affluent. The effect on middle class is mostly insignificant, and no general conclusions can be drawn on ethnic origin as a middle class determinant.

Greater impacts are displayed in Bolivia, Mexico and Peru while the less consistent is in Chile and Colombia.

While the variable is not available in Argentina, neither for most other countries for the period 2000-2003 (even 2006 for Mexico), drawing time evolution of this effect is challenging. We nevertheless note that there is a decreasing impact over the past decade in the countries where this information is available. In Bolivia, Chile, Colombia and Peru, the impact is almost halved over the period. This can be attributed to the fact that along the years, belonging to an ethnic group is less and less a burden on the labor market.

<u>Geographical localization</u>: Living in the capital region is a determinant of higher incomes, while living in the rest of the country is rather a poor and a middle class determinant.

In all countries, except in Bolivia, living in the capital (rather than in the rest of the provinces or regions) displays a positive impact on income (annex 3F), decreasing the likelihood of being poor by -3.1% (Chile) to -11.3% (Peru). It decreases the likelihood of being middle class by -2.1% (Argentina) to -8.1% (Peru), and capital inhabitants are 3.4% (Mexico) to 19.3% more likely to be affluent (Peru). This pattern is exactly the opposite in Bolivia, where Sucre inhabitants are more likely to be poor (+10.3%), and less likely to be affluent (-13.5%).

We observe a decreasing impact along the years in Mexico and Peru. There is no clear tendency in the remaining countries. Indeed, in Argentina, the impact decreases along the years regarding the poor and the middle sector but increases regarding the affluent. In Chile, the impact decreases along the years regarding the poor but increases regarding the middle sectors and the affluent. Finally in Colombia, the impact increases along the years regarding the poor and the affluent but decreases regarding the middle sector.

<u>Activity status and sector of heads of household:</u> Heads of household working in agriculture, being unemployed or inactive are determinants of poverty, while their working in services tends to be more a middle and affluent classes' determinant (annex 3G).

In 2009-2010, unemployed heads of household are more likely to be poor, less likely to be middle class or affluent. Being unemployed displays the highest negative impacts in Bolivia and Chile and the slightest ones in Mexico and Argentina. We generally observe a decreasing tendency of unemployment impact.

Inactive heads of household are more likely to be poor in 2009-2010, and less likely to be middle class. All countries follow the same pattern, except Argentina where inactive effect is positive on the likelihood of being middle class, but not always significantly. Finally in all countries being inactive decreases the likelihood of being affluent.

For active occupied heads of household, we focus on the sector of activity, in comparison with the reference population: working in other services (all services except trade, public and social services).

For heads of household, working in the agricultural sector always displays a strong negative impact on income classes belonging. No clear pattern can be identified of all other sectors: coefficients are less significant and display no homogenous impact in the countries of the sample.

In 2009-2010, working in the agriculture increases the likelihood of being poor and decreases the likelihood of belonging to the middle sectors or affluent. Working in agriculture is therefore a poverty determinant, in almost all Latin American countries. The only noticeable exceptions are 1) Chile, where middle class is also composed of a relatively important proportion of agricultural workers: and 2) Argentina where the effect of working in agriculture is not significant on the affluent side. This may be due to the high value and industrialization of the agricultural sector in Chile and Argentina, wine production, and extensive high-valued production, compared to the rest of the countries studied.

Working in the agricultural sector displays highest consistent negative impact in Bolivia where it represents a heavy burden on income classes belonging and in a lesser extent in Peru. The smallest negative impacts are observed in Chile and Argentina.

The impact is decreasing over the past decade in almost all countries, the exception being made of Peru (for all social classes) where working in agriculture explains more and more the gap between poverty on one hand and middle and affluent class on the other hand.

No clear pattern can be observed in what regards industry workers in Latin America, the only noticeable exception being Chile, where industry occupation is a middle class determinant. In Peru on the other hand, industry occupation is more a poverty determinant.

No clear pattern can be observed for trade workers: The only observable effect is displayed in Bolivia, where trade occupation is an affluent determinant.

Working in public and social services displays two opposite impacts in Latin American countries: in Mexico, Colombia and in a lesser extent Argentina the impact is positive, public and social services occupation protecting individuals from poverty and even from being middle class, and public workers more concentrated in the affluent class. The conclusion is radically different in Chile and (less significantly) in Peru, as the observed impact of public and social services occupation is negative. The effect is almost always insignificant in Bolivia compared to other services.

Public services occupation is therefore a poverty determinant in Chile and Peru, while it is more an affluent determinant in Mexico, Argentina and Colombia.

<u>Heads of household being self-employed:</u> Self-employment is a determinant of poverty, and non self-employment a determinant of middle and affluent class, with the noticeable exception of Chile.

Self-employment displays a negative impact on income classes belonging in all countries except in Chile where it astoundingly displays a positive effect (annex 3H). Self-employment includes a large range of realities, businesses, jobs, and therefore does not seem to represent the same reality across the different Latin American countries of the sample.

In 2009-2010, the head of household being self-employed increases the likelihood of being poor, decreases the likelihood of belonging to the middle sectors (not significant for all countries) or the affluent class (not significant in Mexico).

Chile left apart, the negative impact on poverty displayed by self-employment of household heads is the most consistent in Bolivia and Colombia. The negative impact on middle sector is more consistent in Bolivia and Mexico than in the other countries, and self-employment favors affluency the most in Colombia and Bolivia. On the contrary, it displays the slightest impact in Mexico and Argentina regarding the poor and the affluent side. We observe a decreasing impact along the years in Argentina, Bolivia, Peru and Colombia since 2003.

In Chile as self-employed heads of household are less concentrated in the poor and middle class compared to all other Latin American countries: in Chile 19.7% of the poor occupied heads of household are self-employed, and respectively 21.1% of the middle class occupied and 28.1% of the affluent occupied heads of household. At the other extreme of self-employment pattern in Latin America, in Bolivia, up to 81.8% of the poor occupied are self-employed, and respectively 43.8% of the middle class occupied and 28.9% of the affluent occupied heads of household.

Logically then, being self-employed in Chile decreases the likelihood of being poor (-4.0%), and to be middle class (-14.7%) while it increases the likelihood of being affluent by 18.7% in 2010. These results for Chile are time consistent (always significant), with increasing effects over the past decade: self-employment is becoming more and more an affluent determinant and less and less a poor or middle class one in Chile, while in all other five countries, self-employment remains the situation of almost half of the poor active occupied heads of household.

Other household members' activity status:

As expected, in all countries, although in different proportions, the number of other occupied members in the household displays a positive effect on income classes belonging while the number of unemployed and inactive members in the household displays a negative one (see annex 4 for details).

2. Gender specific effects

The proportion of female headed households is very heterogeneous among Latin American countries ranging from less than 25% in Bolivia, Mexico and Peru to Argentina, over to 35%. Women are also quite well represented in Chile and Colombia as 33% of total households have female heads (table 2).

All classes	Argentina	Bolivia	Chile	Colombia	Mexico	Peru			
male	65,0	75,5	66,9	67,3	75,4	75,0			
female	35,0	24,5	33,1	32,7	24,6	25,0			

Gender of heads of household (2009/2010)

Table 2: Gender of heads of household (2009/2010)

In all countries the wealthier is the household the lower the proportion of female heads, except in Bolivia and Mexico (table 3). The proportion of female headed household is the most important among poor and middle sector households in Argentina and Chile (respectively more than 40% of poor households and more than 30% of middle sector households) while the proportion of female headed households is the most important among affluent Argentinean and Colombian households (around 30%). Higher gaps between affluent and poor households are observed in Chile and in a lesser extent in Argentina.

Poor	Argentina	Bolivia	Chile	Colombia	Mexico	Peru
male	58,5	77,9	55,5	62,5	78,5	74,6
female	41,5	22,1	44,5	37,5	21,5	25,4
Middle sector	Argentina	Bolivia	Chile	Colombia	Mexico	Peru
male	64,6	73,8	66,5	67,8	75,5	74,2
female	35,4	26,2	33,5	32,2	24,5	25,8
Affluent	Argentina	Bolivia	Chile	Colombia	Mexico	Peru
male	69,4	76,0	74,2	70,2	73,4	76,7
female	30,6	24,0	25,8	29,8	26,6	23,3

Gender of heads of household (2009/2010), by classes

Table 3: Gender of heads of household, by income class (2009/2010)

Annexes 5A and 5B present all marginal effects of model 2 multinomial probits for all 6 countries in 2009-2010. Model 2 is similar to model 1, but authorizes specific gender effects of all individual and employment characteristics.

First, it is interesting to compare gender effects on income classes in model 1 and model 2 in the six countries of our panel. Indeed, once controlled for gender specific effects of each individual and employment characteristics, we observe that the gender gap observed previously on class belonging (model 1) is even higher when considering specific gender effects for individual and employment characteristics (model 2).

Age displays relatively few specific gender effects. Indeed, for less than 30 years old, being young is less penalizing for women in Argentina (-12.5pp less likely to be middle sectors and +12.4pp more likely to be affluent), on the contrary it negatively impacts females more than males

in Chile. In what regards heads of households older than 65 years old, there is a significant gender difference in Bolivia, Colombia, Mexico and Peru. Elderly heads positively impacts females more than males in Colombia and Peru. The highest impact is displayed in Bolivia since females have - 17pp chances less than males to belong to the middle sectors, while the slightest impact is displayed in Colombia where female older than 65 years old are -4.2pp less likely than males to be poor.

	up to 3 o	0 years Id	31-40 years old		41-65 years	more f year	than 65 s old	
	М	F	М	F	old	М	F	
ARGENTINA		-	-		Ref.	+		
BOLIVIA	-		-		Ref.	+	less MC	
CHILE	-		-		Ref.	+		
COLOMBIA	-		-	МС	Ref.	+	++	
MEXICO	-		-		Ref.	+	++	
PERU	-		-		Ref.	++	+	
Legend:		Female e	ffect is no	t differen	t from ma	le effect		
	MC	Concentr	ated in m	iddle sect	ors			
	less MC	Concentr	ated in po	oor and af	fluent clas	SS		
	-	Negative	Negatively impacts income classes					
	+	Positively	impacts	income cla	asses			

In all countries, except in Peru, tertiary education displays a positive gender impact. In Argentina, Colombia and Chile tertiary education mostly shelters female from being poor (around - 4pp less likely to be poor than males with the same level of education). In Mexico, female heads of household with tertiary education are -6.1pp less likely to belong to the middle class than males and +9pp more likely to be affluent than males. In Bolivia female heads with tertiary education are more likely to be affluent than their male counterparts. In Colombia secondary educated women are +6.2pp more likely to be affluent than men. In Peru we observe no specific gender.

	Primary	Seco educ	ndary ation	Univ. or educ	tertiary ation
_	education	М	F	М	F
ARGENTINA	Ref.	++	+	+	MC
BOLIVIA	Ref.	+		+	++
CHILE	Ref.	+		+	++
COLOMBIA	Ref.	+	++	++	+
MEXICO	Ref.	+		+	++
PERU	Ref.	+		+	

	Single	Cοι	ıple
		М	F
ARGENTINA	Ref.	MC	+
BOLIVIA	Ref.	-	+
CHILE	Ref.	MC	+
COLOMBIA	Ref.	MC	+
MEXICO	Ref.	-	+
PERU	Ref.	-	+

In all six countries being in couple displays a positive gender impact on females. The highest impact is displayed in Bolivia (-10.7pp gender difference on the likelihood of being poor, -22.7pp gender difference on the likelihood of belonging to the middle class and +33.4pp on the likelihood of being affluent) while the smallest one is displayed in Mexico (-6.6pp difference between male and female heads of household on the likelihood of being poor) and in Argentina (-6.0pp difference on the likelihood of being affluent).

Belonging to an ethnic minority group only displays a specific gender impact in Chile and Bolivia, but not in the same way. In Chile, belonging to an ethnic group disadvantages females more than males (only significant regarding the poor). Indeed it makes females 7.4% more likely to be poor and males only 3.6% more likely. In Bolivia, female heads belonging to an ethnic minority group are less disadvantaged than male heads: they are 2.7% more likely to be poor and male heads 10.0% more likely.

	Etnic majority	Ethnic I gro	minority oup
	group	М	F
ARGENTINA	Ref.	N.A.	N.A.
BOLIVIA	Ref.	-	-
CHILE	Ref.	-	
COLOMBIA	Ref.	-	
MEXICO	Ref.	-	
PERU	Ref.	-	

	Other o mem	ccupied nbers	Other un mem	nemployed Other in mbers memb		nactive 1bers
	М	F	М	F	М	F
ARGENTINA	+	++	-		-	less MC
BOLIVIA	+	++	-		-	
CHILE	+	++	-			-
COLOMBIA	+		-		-	
MEXICO	+		-		-	
PERU	+	++	-		-	

The number of other occupied members displays a positive impact that favors females more than males in almost all countries. The number of unemployed members in the household only displays a gender specific impact in Argentina where it negatively impacts females more than males.

	Agric	ulture	Indu	ıstry	Tra	ade	Public a serv	nd social ⁄ices	Other	Unemp	loyment	Inact	tivity
	М	F	М	F	М	F	М	F	services	М	F	М	F
ARGENTINA	-				-	+	+	-	Ref.	-	MC	-	
BOLIVIA		-							Ref.	-	MC		-
CHILE	-		-		-		-		Ref.	-		-	
COLOMBIA		-					++	+	Ref.	-			-
MEXICO	-		MC+	MC++		-	+	MC	Ref.	-	МС	-	
PERU	-		-		-	0	-	+	Ref.		-		-

: Female effect is not different from male effect

As regards sectors of employment, results vary from country to country:

- Working in agriculture is less penalizing for females than for males in Colombia and Bolivia but negatively impacts females more than males in Mexico.
- Working in industry is more penalizing for females in Chile and polarizes female headed households on middle sector in Mexico.
- Working in trade is less penalizing in Argentina and Peru while in Mexico; it displays no significant impact on males but penalizes women.
- Working in public and social services is more penalizing or less advantaging for females in Argentina, Chile, Colombia and Mexico but displays a positive effect on Peruvian women.
- Being unemployed displays a negative impact on male heads of household in all six countries of our panel. Unemployed female heads of household are more concentrated in middle class than their male counterparts.
- Being inactive displays a negative impact on male heads of household in all six countries, but is less penalizing for female heads in Bolivia, Colombia and Peru. On the contrary female heads of household are more penalized in Mexico.

As seen previously, being self-employed is penalizing for males in all countries except Chile, but is a little less penalizing for females in Colombia and Peru. The notable exception of Chile shows that in this country, being self-employed displays a positive impact on income classes belonging. Though being self-employed is a little less a positive characteristic for females than it is for males.

	No selfemplo	Selfem	ployed	
	yed	М	F	
ARGENTINA	Ref.	-		
BOLIVIA	Ref.	-		
CHILE	Ref.	++	+	
COLOMBIA	Ref.	-	MC	
MEXICO	Ref.	-		
PERU	Ref.		-	

3. <u>Multinomial probit of the spouses' likelihood of being poor, middle sector or affluent</u>

Since more than 70% of the households are headed by males, and more than 80% of which are in couple, we decided to focus on these male headed households including spouses' characteristics as well. It is very interesting to compare the husbands and spouses work status so as to understand how spouses contribute to the household's income (table 4). In Peru and in Bolivia a large amount of spouses are active (70% in Peru and 64% in Bolivia), and therefore not much of them are inactive or unemployed. However Peruvian spouses display the highest level of unemployment of the sample we study (12.6%). In all remaining four countries, spouses are mostly inactive (the highest level being in Chile and Mexico, around 62%).

		-				
SPOUSES	Argentina	Bolivia	Chile	Colombia	Mexico	Peru
Active occupied	45,8	63,9	34,6	44,8	35,0	70,5
Active unemployed	3,0	5,2	3,4	2,1	0,5	12,6
Inactive	51,2	30,8	62,0	53,1	61,5	16,9
Agriculture*	0,4	35,9	7,0	5,8	6,2	32,1
Industry*	7,8	9,9	7,8	14,3	13,9	9,6
Trade*	0,6	23,4	26,0	25,1	27,0	26,2
Publi and social services*	49,7	16,3	47,2	27,2	45,0	14,3
Other services*	41,5	14,5	12,0	27,6	7,8	17,9
Selfemployed*	17,8	33,7	22,3	45,5	27,5	40,8

Activity and employment characteristics of spouses and husbands in households

* For active occupied spouses only

HUSBANDS	Argentina	Bolivia	Chile	Colombia	Mexico	Peru
Active occupied	80,6	94,0	79,2	85,6	83,9	90,5
Active unemployed	2,9	1,6	3,2	7,4	2,8	4,0
Inactive	16,4	4,4	17,6	6,8	11,1	5,5
Agriculture*	1,5	31,5	14,8	24,8	20,2	32,8
Industry*	17,9	15,6	15,9	12,7	15,8	12,5
Trade*	14,6	10,3	15,5	20,6	14,0	13,1
Publi and social services*	10,9	12,3	18,1	10,0	21,5	14,6
Other services*	55,1	30,3	35,6	32,0	28,5	27,0
Selfemployed*	22,2	45,6	22,3	48,6	26,9	44,6

* For active occupied husbands only

Table 4: Activity and employment status of spouses and husbands (2009/2010)

Model 3 is based on a different population than models 1 and 2, and focuses on male headed households with spouses. The aim of this last model is to identify the relative importance of spouses' decisions and characteristics compared to their husband's ones. As presented in section IV, the empirical framework is a multinomial probit estimation on spouses only, including both their own individual and work characteristics and their husband's characteristics. Annex 6 presents model 3 multinomial probit estimations (marginal effects), in all 6 countries for 2009-2010.

					2009 / 2010
	VARIABLES	Poor	Middle class	Affluent	Exceptions / notes
	up to 30 years old 31-40 years old 41-65 years old more than 65 years old	+ + Ref. -	+ / n.s. + Ref. - / n.s.	- - Ref. +	Colombia, elderly are not significantly richer
	Primary education Secondary education Technical education or University	Ref.	Ref.	Ref. + +	Bolivia, secondary education decreases poverty and tertiary education increases affluency, other effects are insignificant.
	Majority group Ethnic group	Ref. +	Ref. n.s.	Ref.	Effect unclear on middle class, variable not available on Argentina
Spouse	Other provinces/regions Capital	Ref.	Ref. n.s.	Ref. +	Bolivia, living in Sucre positively impacts poverty and negatively impacts affluency
caracteristics	Spouse active occupied, agriculture	+	n.s.	-	Except in Argentina, spouses in agriculture lower the likelyhood to be poor
	Spouse active occupied, industry	+	variable	-	Except Colombia
	Spouse active occupied, trade	Mostly ins ad	signficant, ar	nd variable ies	Negative impact on income class in Chile and Mexico
	Spouse active occupied, public and social services	Mo	stly insignfic	cant	Negative impact on income class in Argentina
	Spouse active occupied, other services		Ref.	Ref.	
	Spouse active unemployed	+	n.s.	-	Observed in all 6 countries
	Spouse inactive	+	+	-	Observed in all 6 countries
	Spouse no self-employed	Ref.	Ref.	Ref.	Chile : polarization on poor and affluent, Bolivia no significant
-	Spouse self-employed	+	+	-	· · · · · · · · · · · · · · · · · · ·
	Husband primary educated	Ref.	Ref.	Ref.	
	Husband secondary educated	-	-	+	Not significant on middle sectors in Bolivia and Colombia
	Husband technical educated or University	-	-	+	Observed in all 6 countries
	Husband active occupied, agriculture	+	-	-	Chile: + on middle class
	Husband active occupied, industry	n.s.	n.s.	n.s.	Argentina and Mexico : - on poor, Chile :+ on middle class and - on affluent
Husband	Husband active occupied, trade	Variab	le across co	ountries	Globally, negative impact on income classes
individual caracteristics	Husband active occupied, public and social services	Variab	le across co	ountries	Positive impact on income classes in Argentina, Colombia and Mexico. Negative one in Chile and Peru
	Husband active occupied, other services	Ref.	Ref.	Ref.	
	Husband active unemployed	+	-	-	Observed in all 6 countries
н	Husband inactive	+	-		+ on middle class in Argentina, not significant in Chile, polarized effect in Mexico
	Husband no self-employed Husband self-employed	Ref. +	Ref.	Ref.	Chile: all the contrary: - on poor and MC, + on affluent

Table 5: Synthesis of marginal effects signs and significance in MNP Model 3

The impact of spouses' age is very similar to the one observed in model 1 performed on heads of household, the younger a spouse is, the more likely her household will be poor and in a lesser extent belong to the middle sector. The older she is, the more likely her household will become affluent. Conclusions are also similar to model 1 in what regards ethnicity and geographic localization.

It is of interest to focus on the impact of spouses' education and employment status on their household income class belonging.

Spouses' education displays a positive impact on income classes belonging in all countries, although with different magnitudes and with a smaller impact than that displayed by husbands' education in all countries except Colombia. Indeed spouses reaching secondary education decreases the household's likelihood of being poor or middle sector and increases the household's likelihood of being affluent. The highest impacts of spouses' secondary education are displayed in Colombia, Mexico and Chile. The smallest ones are displayed in Peru and Bolivia (and Mexico regarding the middle sector).

Spouses reaching tertiary education do not have a significant impact in Bolivia, but decrease the household's likelihood of being poor in all other countries or middle sector and increases their likelihood of belonging to the affluent class. The highest impacts of spouses' tertiary education are displayed in Colombia and Chile regarding the middle sector and the affluent and in Argentina and Mexico regarding the poor. The smallest impacts are displayed in Bolivia and Peru.

Working in the agricultural sector displays a negative impact on spouses in all countries except in Argentina where the impact is positive and in Colombia where it is insignificant. Spouses working in the agricultural sector increase the household's likelihood of being poor or middle sector and decreases the likelihood of being affluent. The highest impacts are displayed in Bolivia and Peru while the smallest are displayed in Chile and Mexico. Interestingly, Argentinean spouses working in agriculture decrease their households' likelihood of being poor by -10%.

The other sectors display different impacts regarding the countries. For instance, working in industry has a relatively constant negative impact on spouses. Working in public and social services displays a negative impact in Argentina and Chile and a positive impact in Bolivia (reducing poverty by -10.7%) but is not significant in the remaining countries. Spouses working in public and social services increase the household likelihood of being poor or middle sector and decrease their likelihood of being affluent.

Spouses being unemployed and inactive display a negative impact in all six countries.

Spouses being unemployed increases the household's likelihood of being poor, and decreases their likelihood of being affluent. There is no impact regarding the middle sector. The highest impacts are displayed in Chile, Bolivia and Argentina while the smallest are displayed in Peru, Mexico and Colombia.

Spouses being inactive increase the household's likelihood of being poor or middle sector (not significant in Bolivia and Peru), and decreases the household's likelihood of belonging to the affluent. The impacts are therefore the more burdensome in Chile and Argentina while they are the less disadvantaging in Peru and Mexico.

Spouses being self-employed display a negative impact in Argentina, Colombia, Mexico and Peru, no impact in Bolivia and a polarized one in Chile (it both increases the household's likelihood of being poor and affluent, respectively by 5.4% and 4.2%, and decreases its likelihood of belonging to the middle sector by -9.6%). Indeed, spouses being self-employed increase the household's likelihood of being poor or middle sectors (not significant in Mexico), and decreases their likelihood of being affluent. The highest impacts are displayed in Colombia and Argentina while the smallest ones are displayed in Peru and Mexico.

In terms of husbands' characteristics impact on household's income classes, as seen in model 1, head of household's education always displays a positive impact on income classes belonging. Marginal effects are nevertheless smaller in model 3 than they were in model 1. Indeed, due to the correlation between husbands' and spouses' education, marginal effects in model 1 are capturing a part of spouses' education.

The highest impacts of both husbands' secondary and tertiary education are displayed in Chile, Argentina and Mexico, while the smallest impacts are displayed in Peru and Bolivia.

Husbands' education impact more income classes than spouses' education in all countries except in Colombia and in a lesser extent Argentina (only on the poor side of probits). No significant difference is observed in Peru.

As seen in model 1, husbands working in agriculture display a negative impact on household's income classes in all six countries, but marginal effects are here again lower in model 3 than in model 1, due to the correlation between husbands' and spouses' sectors of activity. The negative impact of agriculture is therefore higher when spouses work in agriculture than when husbands do in Bolivia, Chile and Peru. There are differences in what regards the impact of other sectors. For instance, husbands working in trade display a negative impact in Argentina and Peru (no impact for spouses working in trade). Husbands working in public or social services display a positive impact in Argentina, Colombia and Mexico, a negative one in Chile and Peru (no impact in Bolivia) while the effect of spouses working in public or social services is negative in Chile and Argentina, positive in Bolivia and not significant in the 3 remaining countries.

Husbands being unemployed or inactive display, as in model 1, a striking negative impact on household's income classes belonging in all countries. Marginal effects of unemployment in model 3 are a little smaller than in model 1. Marginal effects of husbands being unemployed are always higher than the ones of spouses being unemployed, the activity status of husbands is therefore more important than the one of their spouses and has a higher impact on household income class. The only two exceptions are Colombia and Peru where spouses being unemployed decrease more the household's likelihood of being affluent than husband's unemployment. Marginal effects of husbands' inactivity in model 3 are smaller than in model 1 except in Bolivia and Mexico. Marginal effects of husbands being inactive are lower than the ones of spouses being inactive in Argentina, Chile and Mexico (Colombia and Peru only on the affluent side of probits), meaning that spouses' inactivity is a striking determinant of income classes belonging. Indeed, the inactivity rate of husbands (it varies from 4.4% to 17.6% across countries) is very low in comparison with spouses' inactivity rate (16.9% to 62%).

In conclusion, female labor market participation is a strong vector of income classes' improvement in Latin American countries, and should be a political economy priority for governments.

Finally, husbands being self-employed display a negative impact in all countries except in Chile and Mexico where it has a polarized effect (both increasing the household's likelihood of being poor and affluent, and decreasing its likelihood of belonging to the middle sector). Marginal effects of husband's self-employment in model 3 are smaller than in model 1 in all countries except in Chile, where self-employment not only displays a positive impact, but has an even higher positive effect once controlled for spouses' characteristics (model 3). Our last specification reinforces this special result for Chile. Marginal effects of husbands' self-employment are higher than the ones of spouses' self-employment in all countries, except in Argentina, Colombia and Mexico where spouses' self-employment decreases in a higher extent the household's likelihood of being affluent.

VI. Conclusion

The study of the characteristics that makes an individual more or less likely to be poor or affluent rather than middle class through the examples of Argentina, Bolivia, Chile, Colombia, Mexico and Peru, show that four main characteristics put middle class households at risk of falling into poverty. They lack education, unemployment, inactivity and self-employment. Furthermore in terms of sector of activity it should be taken into account that the agricultural sector is a strong vector of poverty and is of relevant importance when poor people mostly on it.

On the contrary, education (secondary or tertiary completed) and employment (depending on the sector) somehow promote the middle class social ascension. Therefore, in terms of public policies the priorities should be to promote schooling such as through good access and infrastructures, or through encouraging poor families who send their children to school instead of having them work (such as it has already been done in Mexico or Brazil) for instance. An emphasis could also be made on fighting unemployment and help individuals remain in the labor market or have access to part-time jobs.

A special effort should be made on promoting and favoring female labor market participation (for example increasing child care facilities such as day care centers so that women would not step out of the labor force for a too long time when having children). In the same way, efforts that have been made in Latin America to promote females' education in the past decades should be fulfilled. Poor and middle class female heads of household could also be supported through social allowances for instance. Finally, self-employed workers should be taken into account in the design of social protection policies, pension policies but not only, so as to provide them a safety net.

However, given the large informality and the heterogeneity of Latin American economies, the policies might not display all the expected results.

VII. References

ALKIRE, S., and J. Foster, 2007, "Counting and multidimensional poverty measurement", Oxford Poverty and Human Development Initiative, Working Paper 7, University of Oxford

ALKIRE, S. and M.E. Santos, 2010a, "Acute multidimensional poverty: a new index for developing countries.", Oxford Poverty and Human Development Initiative, Working Paper 38, University of Oxford

AMORANTE, G., N. Chun, and A. Deolalikar, 2010, "Who are the middle class and what valued do they hold? Evidence from the world values survey", Working Paper 229, Asian Development Bank

ATAL, V., 2010, "A theory of female labor supply", Working Paper, Department of Economics, Cornell University

ATAL, J.P., H. Nopo and N. Winder, 2009, "New century, old disparities. Gender and ethnic wage gaps in Latin America", Working Paper No 109, Inter-American Development Bank, Research Department

ATTANASIO, O. and M. SZékely, 2002, "A dynamic analysis of household decision-making in Latin America: changes in household structure, female labor force participation, human capital and its returns", Working Paper No R-452, Inter-American Development Bank, Research Department

BADEL, A., X. Pena, "Decomposing the gender wage gap with sample selection adjustment: Evidence from Colombia", working paper, Federal Reserve Bank of St Louis and Universidad de Los Andes

BANERJEE, A. and E. DUFLO,2008, "What is Middle Class about the Middle Classes around the World?", Journal of Economic Perspectives, Vol. 22, No. 2, Spring 2008, pp. 3–28.

BIRDSALL, N., 2007, "Stuck in the Tunnel: Is Globalization Muddling the Middle Class?"Center for Global Development Working paper N° 130 $\,$

BIRDSALL, N.,2010, "The (Indispensable) Middle Class in Developing Countries; or The Rich and the Rest, Not the Poor and the Rest", Working Paper 207, Center for Global Development, Washington, DC.

BIRDSALL, N., C. GRAHAM, and S. PETTINATO, 2000, "Is Globalization Muddling the Middle Class?", Brookings Institution Center Working Paper N°14. Washington, D.C.

BLAU, D.J., et M.L. Khan, 2000, "Understanding international differences in the gender pay gap", Working Paper N° 8200, National Bureau of Economic Research, Cambridge, MA

BOURGUIGON F., and S. Chakravarty, 2003, "The measurement of multidimensional poverty", Journal of Economic Inequality, Springer. Vol. 1(1), pp 25-49, April

BRUCK, T., A.M.DANZER, A. MURAVYEV and N.WEISSHAAR, 2007, « Determinants of poverty during transition: household survey evidence from Ukraine », PRUS working paper N°40, Ukraine

CALONICO, S., and H. Nopo, 2009, "Gender segregation in the workplace and wage gap: evidence from urban Mexico 1994-2004" in *Research on Economic Inequality*. Vol 17 Editor(s): Yves Fluckiger, Sean F. Reardon, Jacques Silber ISBN: 978-1-84855-786-4

CARDENAS, M., Kharas, H. and C. Henao, 2011, "Latin America's global middle class", Working Paper, Global Economy and Development at Brookings, Whashington, D.C.

CRUCES, G., L. Lopez-Calva and D. Battiston, 2011, "Down and out or up and in ¿ Polarization based measures of the middle class in Latin America.", CEDLAS, Working Paper 0113, CEDLAS, Universidad Nacional de La Plata

CASTELLANI, F., G. PARENT, 2010, "Being Middle Class in Latin America", mimeo, OECD development Centre, Paris.

CONNELLY, R., D. De Graff, and D. Levinson, 1996, "Women's employment and child care in Brazil." Economic Development and Cultural Change 44: 619-56.

DAUDE, C., 2011, "Ascendance by Descendants? On Intergenerational Education Mobility in Latin America", , OECD Development Centre Working Paper No. 297, Paris.

DE LA RICA, S., J. Dolado, V; Llorens, 2005, "Ceiling and floors: gender wage gaps by education in Spain", IZA Discussion papers 1483, Institute for the Study of Labor (IZA)

DURYEA, S., S. Galiani, H. Nopo and C. Piras, 2007, "The educational gender gap in Latin America an d the Caribbean", Working Paper No 600, Inter-American Development Bank, Research Department

EASTERLY, W., 2001 "The middle class consensus and economic development", Journal of Economic Growth, vol. 6(4), pp. 317-35

ECLAC 2009, 2010, Anuario Estadístico de America Latina y el Caribe, United Nations Economic Commission for Latin America and the Caribbean, Santiago de Chile.

EISENHAUER, J.G.,2008, "An Economic Definition of the Middle Class", Forum for Social Economics, Vol. 37, No. 2, pp. 103-113.

FIESS N.M. and D. VERNER, 2004, « The dynamics of poverty and its determinants : the case of the Northeast of Brazil and its states », Policy Research working paper N°3259, World Bank, Washington D.C.

FAJARDO, J. and E. LORA,2010, Understanding the Latin American Middle Classes: Reality and Perception", mimeo,Inter-American Development Bank, Research Department, June.

GALVIS, L.A., 2010, "Diferenciales salariales por género y region en Colombia: una aproximacion con regresion por cuantiles", Documento de Trabajo sobre economia regional No 131, Centre de Estudios Economicos Regionales, Banco de la Republica, Cartagena

GASPARINI, L., S. Galiani, G., Cruces and A. Pablo, 2011, "Educational upgrading and returns to skills in Latin America Evidence from a supply-demand framework, 1990-2010", IZA Discussion papers 6244, Institute for the Study of Labor (IZA)

HAUSMANN R., L.D. Tyson and S. Zahidi, 2010, "The global gender gap report", World Economic Forum. Geneva, Sxwitzerland

KANDIL L. E., 2009, "Gender wage discrimination in Egypt: a quantile regression analysis", Paris: Université Paris I Panthéon-Sorbonne

KHARAS, H., and G. Gertz (2010), The Emerging Middle Class in Developing Countries, OECD Development Centre Working Paper No. 285, Paris.

KHARAS, H., and H. Kholi, 2011, "What is the middle income trap. Why do countries fall into it and how can it be avoided ?", Global Journal of Emerging Market Economies, 3 pp. 281-289

LOPEZ-CALVA, L.F. and N. LUSTIG,2010, Declining Inequality in Latin America : A Decade of Progress (New York: United Nations Development Programme, and Washington DC: Brookings Institution.

LOPEZ-CALVA, L.F., and E. Ortiz-Juarez, 2011, "A vulnerability approach to the definition of the middle class.", Policy research working paper series 5902, The World Bank, Washington D.C.

LOPEZ-CALVA, L.F., J. Rigolini and F. Torche, 2011, "Is there such thing as middle class values? Class differences, values and political orientations in Latin America.", Policy Research Working Paper Series 5874, The World Bank, Washington D.C.

LUSTIG, N., L.F. Lopez-Calva and E. Ortiz-Juarez, 2011, "The decline in inequality in Latin America: How much, since when and why", working paper 1118, Tulane University, Department of Economics

MARSHALL, J., and V. Calderon, 2005, "Social exclusion in Education in Latin America and the Caribbean", Discussion Draft, Inter-American Development Bank

MEYER, B.D. and D.T. Rosenbaum, 2001, "Welfare, the earned income tax credit, and the labor supply of single mothers", Quaterly Journal of Economics, 116, 1063-1114

MIZALA, A., P. Romaguera, P. Henriquez, 1999, "Female labor supply in Chile", Working Paper, Centro de Economia Aplicada, Universidad de Chile

NOPO, H., N. Daza and J. Ramos, 2011, "Gender earning gaps in the world", IZA discussion papers 5736, Institute for the study of labor (IZA)

OECD, 2010, Latin American Economic Outlook 2011, OECD Development Centre, Paris

PRESSMAN, S. 2007, "The decline of the middle class: an international perspective," Journal of Economic Issues, 40 (1).

RAVALLION, M. 2009, "The Developing World's Bulging (but Vulnerable) Middle Class", World Development, Vol. 38, No. 4, pp. 445-454.

RAVALLION, M., 2011, "On multidimensional indices of poverty." Policy research working paper series 5580, The World Bank, Washington D.C.

SOLIMANO, A.,2008, "The Middle Class and the Development Process", Serie Macroeconomía del Desarrollo No. 65. Santiago, Chile: United Nations-ECLAC.

SCHONBERG, U., and J. Ludsteck, 2007, "Maternity leave legislation, female labor supply and the family gap", IZA discussion papers 2699, Institute for the study of labor (IZA)

THUROW, L.C., 1984, "The Disappearance of the Middle Class," New York Times, 3, 2.

TORCHE, F. and L.F. LOPEZ-CALVA (2010). "Stability and Vulnerability of the Latin American Middle Class", forthcoming in K. Newman ed. Dilemmas of the Middle Class around the World. New York: Oxford University Press.

WONG, R. and R.E. Levine, 1992, "The effect of household structure on women's economic activity and fertility: evidence from recent mothers in urban Mexico", Economic Development and Cultural Change, 41(1), p. 89-102

WORLD BANK, 2005, *Introduction to poverty analysis*, World Bank, Washington D.C.

WORLD BANK, 2011, "A break with history: fifteen years of inequality reduction in Latin America" Brief, *Poverty and Gender Unit from the Poverty Reduction and Economic Management Team (LCSPP) in the Latin America an Caribbean Region*

VIII. Annexes

						Propo	rtion of total hous	eholds
Country	Conducted by	Type of survey	Survey	Year	Sample size (number of households)	Poor	Middle class	Affluent
	INDEC (Instituto		EPH - Encuesta Permanente de	2000	22487	22,5	45,5	32,0
Argentina	Nacional de Estadística y	household	Hogares	2003	25838	28,8	37,2	34,0
Aigentina	Censos)	survey	EPHC - Encuesta permanente de	2006	37521	23,8	43,3	33,0
			hogares continua	2010	43610	17,9	52,0	30,2
Polivio	INE (Instituto nacional de	household	ECH - Encuesta continua de hogares-condición de vida	2000	4857	31,7	33,1	35,3
DOIIVIA	estadísticas)	survey	Ell. Enguesto de bagaros	2005	4086	27,6	36,7	35,7
			EH - Encuesta de nogares	2009	4034	23,1	44,9	32,0
			CASEN Enguesta da	2000	65036	19,9	47,1	33,0
Chile		household	CASEN - Elícuesia de	2003	68153	19,1	48,5	32,5
Chile	MIDEPLAN	survey		2006	73720	18,7	49,1	32,2
			INACIONAI	2009	71460	17,6	50,4	32,0
-	DANE (Departamento			2003	22949	37,0	24,3	38,7
Colombia	Administrativo Nacional	LSMS	ECV - Encuesta de calidad de vida	2008	13595	25,8	39,5	34,7
	de Estadistica)			2010	14787	20,3	46,9	32,8
	INECI (Institute pasional			2000	10108	20,0	48,0	32,0
Maxiaa	do ostadística v	household	ENIGH - Encuesta nacional de	2004	22595	17,2	52,0	30,7
Mexico		survey	ingresos y gastos en los hogares	2006	20875	17,1	52,9	30,0
	geografia)			2010	27655	16,8	53,4	29,8
	INEL (instituto pacional		ENAHO - Encuesta nacional de	2000	3721	23,6	45,8	30,6
Poru	do ostadísticas o	household survey p	old hogares - Condiciones de vida y	2003	12580	21,7	46,0	32,3
Peru de in	informática)			2006	20577	21,6	47,3	31,2
			p00162a	2010	21496	20,0	50,7	29,3

Annex 1: Data presentation by country: survey, sample size, proportions of poor, middle class and affluent households

		ARGENTIN	Α	BOLIVIA		CHILE		COLOMBIA			MEXICO			PERU				
		2010			2009			2009			2010			2010			2010	
VARIABLES	Poor	Middle	Affluont	Poor	Middle	Affluont	Poor	Middle	Affluont	Poor	Middle	Affluont	Poor	Middle	Affluont	Poor	Middle	Affluont
VANABLEO	1001	sectors	Amuent	FUUI	sectors	Annuent	FUUI	sectors	Annuent	1001	sectors	Amuent	FUUI	sectors	Annuent	1001	sectors	Annuent
Mala	Def	Def	Def	Def	Def	Def	Def	Def	Def	Def	Def	Def	Def	Def	Def	Def	Def	Def
Male	Ref.	Ker. 0.033***	-0.003***	-0.021	Ker. 0.006***	Ref. -0.075***	Ref.	Ref. 0.040***	Ref. -0.087***	Ref.	Ref. 0.054***	Rer. -0 11/***	Ref.	Ref. 0.006	-0.050***	Ref.	Ker. 0.037**	Ref.
1 emaie	(0.000	(0.033	(0.093	(0.021)	(0.030	(0.028)	(0.047	(0.040	(0.011)	(0.000	(0.034	(0.015)	(0.044	(0.014)	(0.030	(0.010)	(0.037	(0.014)
up to 30 years old	0.134***	0.016	-0.150***	0.021)	0.108***	-0 119***	0.053***	0.038***	-0.092***	0.054***	0.109***	-0.163***	0.157***	0.029**	-0 186***	0.070***	0.046***	-0.116***
	(0.013)	(0.013)	(0,009)	(0.024)	(0.028)	(0.025)	(0.010)	(0.014)	(0.013)	(0.015)	(0.017)	(0.014)	(0.014)	(0.014)	(0.009)	(0.015)	(0.017)	(0.014)
31-40 years old	0.102***	-0.010	-0.092***	0.003	0.085***	-0.088***	0.037***	0.039***	-0.076***	0.023**	0.110***	-0.133***	0.097***	0.054***	-0.151***	0.058***	0.034***	-0.093***
	(0.010)	(0.011)	(0.009)	(0.021)	(0.026)	(0.023)	(0.007)	(0.010)	(0.010)	(0.011)	(0.015)	(0.014)	(0.010)	(0.011)	(0.009)	(0.010)	(0.012)	(0.011)
41-65 years old	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
more than 65 years old	-0.111***	0.037***	0.074***	-0.056***	-0.001	0.058	-0.077***	-0.027**	0.104***	-0.040***	-0.054***	0.094***	-0.026***	-0.033**	0.059***	-0.031***	-0.062***	0.093***
	(0.006)	(0.013)	(0.014)	(0.020)	(0.035)	(0.035)	(0.004)	(0.013)	(0.015)	(0.011)	(0.020)	(0.021)	(0.008)	(0.014)	(0.014)	(0.008)	(0.015)	(0.016)
Primary education	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
Secondary education	-0.094***	-0.062***	0.156***	-0.090***	-0.039	0.129***	-0.096***	-0.120***	0.216***	-0.128***	-0.075***	0.203***	-0.106***	-0.119***	0.225***	-0.086***	-0.076***	0.162***
	(0.006)	(0.010)	(0.010)	(0.016)	(0.026)	(0.026)	(0.004)	(0.009)	(0.010)	(0.008)	(0.014)	(0.015)	(0.005)	(0.012)	(0.012)	(0.006)	(0.013)	(0.013)
Technical education or University	-0.196***	-0.243***	0.440***	-0.144***	-0.198***	0.342***	-0.149***	-0.466***	0.615***	-0.181***	-0.377***	0.559***	-0.169***	-0.357***	0.526***	-0.155***	-0.271***	0.427***
	(0.005)	(0.011)	(0.010)	(0.018)	(0.028)	(0.028)	(0.004)	(0.010)	(0.010)	(0.007)	(0.016)	(0.016)	(0.005)	(0.011)	(0.011)	(0.006)	(0.015)	(0.015)
Single	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
Couple	-0.041***	0.024**	0.017*	0.031	0.064**	-0.095***	-0.029***	0.011	0.018*	-0.016	0.034**	-0.017	0.023***	0.046***	-0.069***	0.014	0.033**	-0.047***
	(0.008)	(0.010)	(0.010)	(0.021)	(0.028)	(0.030)	(0.006)	(0.010)	(0.011)	(0.011)	(0.016)	(0.016)	(0.008)	(0.014)	(0.013)	(0.009)	(0.014)	(0.015)
Majority group				Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
Ethnic group	N.A.	N.A.	N.A.	0.076***	0.040*	-0.116***	0.048***	-0.015	-0.033**	0.041***	-0.029*	-0.013	0.084***	-0.014	-0.071***	0.068***	0.011	-0.079***
				(0.016)	(0.021)	(0.020)	(0.008)	(0.014)	(0.015)	(0.012)	(0.017)	(0.018)	(0.007)	(0.010)	(0.009)	(0.007)	(0.010)	(0.009)
Other provinces/regions	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
Capital	-0.053***	-0.021***	0.074***	0.103***	0.033	-0.135***	-0.031***	-0.039***	0.070***	-0.097***	-0.046***	0.143***	-0.045***	0.012	0.034***	-0.113***	-0.081***	0.193***
	(0.005)	(0.007)	(0.007)	(0.020)	(0.022)	(0.020)	(0.005)	(0.008)	(0.009)	(0.011)	(0.018)	(0.018)	(0.006)	(0.009)	(0.009)	(0.008)	(0.012)	(0.012)
Number of other occupied household members	-0.065^^^	-0.002	0.066***	-0.049^^^	-0.012	0.061	-0.136***	0.005	0.130***	-0.092***	-0.001	0.093***	-0.044***	0.002	0.042***	-0.036***	-0.005	0.041
Number of other upemployed boundhold members	(0.004)	(0.005)	(0.005)	(0.009)	(0.010)	(0.010)	(0.004)	(0.005)	(0.005)	(0.005)	(0.007)	(0.007)	(0.003)	(0.005)	(0.005)	(0.003)	(0.004)	(0.004)
Number of other unemployed household members	(0.094	0.060	-0.154	0.005	0.065	-0.070	(0.005)	0.050	-0.131	(0.057	0.035	-0.092	0.059	0.035	-0.093	0.023	(0.001)	-0.074
Number of other inactive bousehold members	(0.006)	(0.015)	0.150***	(0.024)	0.046***	(0.030)	(0.005)	0.046***	0.001***	(0.012)	(0.019)	0.115***	(0.010)	0.016)	0.000***	(0.006)	(0.009)	0.046***
Number of other mactive household members	(0.003)	(0.005)	-0.150	(0.024	(0.040	-0.070	(0.040	(0.040	-0.094	(0.029	(0.000)	-0.115	(0.045	(0.045)	-0.090	(0.024	(0.023	-0.040
Active occupied agriculture	0.122***	-0.096**	-0.026	0.365***	-0 121***	-0 244***	0.068***	0.048***	-0 117***	0.203***	-0.050***	-0 154***	0.233***	-0.102***	-0 131***	0.288***	-0.119***	-0.169***
Active occupied, agriculture	(0.034)	(0.039)	(0.020	(0.029)	(0.030)	(0.022)	(0,009)	(0.011)	(0.010)	(0.017)	(0.030	(0.016)	(0.016)	(0.017)	(0.014)	(0.015)	(0.015)	(0.012)
Active occupied industry	-0.016	0.020	-0.004	0.039	-0.007	-0.032	0.013	0.067***	-0.080***	0.009	-0.022	0.013	-0.020**	0.059***	-0.038***	0.058***	-0.036*	-0.022
	(0.012)	(0.015)	(0.013)	(0.031)	(0.034)	(0.030)	(0.010)	(0.014)	(0.013)	(0.018)	(0.022)	(0.022)	(0.010)	(0.016)	(0.015)	(0.017)	(0.019)	(0.016)
Active occupied, trade	0.085***	0.023	-0.107***	0.014	-0.072**	0.059*	0.002	0.048***	-0.050***	-0.011	-0.013	0.024	-0.022**	0.040***	-0.018	0.031**	-0.014	-0.017
	(0.013)	(0.016)	(0.013)	(0.031)	(0.035)	(0.035)	(0.009)	(0.013)	(0.012)	(0.015)	(0.020)	(0.020)	(0.010)	(0.015)	(0.014)	(0.015)	(0.017)	(0.016)
Active occupied, public and social services	0.022**	-0.029**	0.006	-0.005	-0.012	0.018	0.015*	0.028**	-0.043***	0.013	-0.106***	0.093***	-0.033***	-0.005	0.038***	0.046**	-0.019	-0.027*
	(0.010)	(0.012)	(0.011)	(0.036)	(0.037)	(0.033)	(0.008)	(0.012)	(0.011)	(0.020)	(0.023)	(0.024)	(0.009)	(0.014)	(0.013)	(0.019)	(0.019)	(0.015)
Active occupied, other services	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
Active unemployed	0.277***	-0.149***	-0.128***	0.475***	-0.189***	-0.286***	0.450***	-0.194***	-0.256***	0.305***	-0.184***	-0.121***	0.246***	-0.089***	-0.158***	0.302***	-0.145***	-0.157***
	(0.027)	(0.025)	(0.020)	(0.067)	(0.065)	(0.016)	(0.021)	(0.021)	(0.010)	(0.031)	(0.028)	(0.022)	(0.030)	(0.029)	(0.019)	(0.029)	(0.027)	(0.015)
Inactive	0.176***	0.013	-0.189***	0.330***	-0.142***	-0.188***	0.220***	-0.006	-0.213***	0.272***	-0.111***	-0.161***	0.045***	-0.058***	0.013	0.138***	-0.095***	-0.043**
	(0.012)	(0.013)	(0.010)	(0.048)	(0.043)	(0.027)	(0.012)	(0.013)	(0.012)	(0.023)	(0.022)	(0.018)	(0.013)	(0.016)	(0.015)	(0.024)	(0.023)	(0.019)
No self employed	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
Selfemployed	0.126***	-0.017	-0.110***	0.193***	-0.047**	-0.146***	-0.040***	-0.147***	0.187***	0.170***	0.003	-0.173***	0.075***	-0.068***	-0.007	0.144***	-0.028**	-0.116***
	(0.011)	(0.012)	(0.009)	(0.020)	(0.024)	(0.021)	(0.005)	(0.010)	(0.011)	(0.011)	(0.014)	(0.013)	(0.008)	(0.011)	(0.011)	(0.008)	(0.011)	(0.011)
Observations	43,609	43,609	43,609	4,006	4,006	4,006	70,702	70,702	70,702	14,531	14,531	14,531	27,652	27,652	27,652	21,495	21,495	21,495

Standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1

Annex 2: Multinomial probit of the likelihood of being poor/middle sectors/affluent (on households) – Model 1

Annex 3.A to 3.H: marginal effects of MNP estimations (Model 1) for 2000-2010

GENDER			Poor				Middle class				Affluent			
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	
Male		Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	
	Argentina	0.036**	0.047***	0.051***	0.060***	0.032*	0.038***	0.023*	0.033***	-0.067***	-0.085***	-0.074***	-0.093***	
	Bolivia	n.s.	n.a.	-0.048**	n.s.	n.s.	n.a.	0.094***	0.096***	n.s.	n.a.	n.s.	-0.075***	
Famala	Chile	0.058***	0.047***	0.055***	0.047***	0.023*	0.045***	0.019*	0.040***	-0.081***	-0.092***	-0.074***	-0.087***	
remaie	Colombia	0.127***	n.s.	0.128***	0.060***	n.s.	0.087***	n.s.	0.054***	-0.126***	-0.073***	-0.150***	-0.114***	
	Mexico	0.052***	0.030**	0.036**	0.044***	n.s.	n.s.	n.s.	n.s.	n.s.	n.s.	-0.044***	-0.050***	
	Peru	n.s.	n.s.	n.s.	n.s.	n.s.	n.s.	n.s.	0.037**	n.s.	n.s.	-0.032**	-0.047***	

Annex 3.A:	Gender	marginal	effects	of MNP -	Model 1	(2000-2010	1
	aonaor		0110000		nouce a		J

AGE			Po	or			Middle	e class			Afflu	uent	
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010
41-65 years old		Ref.											
	Argentina	n.s.	n.s.	n.s.	0.134***	0.118***	0.091***	0.081***	n.s.	-0.139***	-0.093***	-0.096***	-0.150***
	Bolivia	n.s.	n.a.	n.s.	n.s.	0.112***	n.a.	0.163***	0.108***	-0.118***	n.a.	-0.174***	-0.119***
up to 30 years	Chile	0.101***	0.134***	0.070***	0.053***	0.074***	0.027*	0.055***	0.038***	-0.175***	-0.162***	-0.125***	-0.092***
old	Colombia	0.075***	0.086***	0.073***	0.054***	0.120***	0.084***	0.080***	0.109***	-0.195***	-0.171***	-0.153***	-0.163***
	Mexico	0.099***	0.085***	0.164***	0.157***	0.095***	0.098***	0.045**	0.029**	-0.194***	-0.183***	-0.209***	-0.186***
	Peru	0.071**	0.064***	0.104***	0.070***	n.s.	0.071***	0.083***	0.046***	-0.110***	-0.136***	-0.187***	-0.116***
	Argentina	0.033**	0.027*	0.027**	0.102***	0.057***	0.048***	0.045***	n.s.	-0.090***	-0.075***	-0.072***	-0.092***
	Bolivia	0.071***	n.a.	n.s.	n.s.	0.058**	n.a.	0.066**	0.085***	-0.129***	n.a.	-0.099***	-0.088***
21-40 years old	Chile	0.103***	0.089***	0.092***	0.037***	0.037***	0.038***	0.023**	0.039***	-0.140***	-0.128***	-0.115***	-0.076***
51-40 years olu	Colombia	0.066***	0.062***	0.046***	0.023**	0.061***	0.062***	0.088***	0.110***	-0.126***	-0.124***	-0.134***	-0.133***
	Mexico	0.106***	0.095***	0.075***	0.097***	0.058***	0.059***	0.063***	0.054***	-0.164***	-0.154***	-0.138***	-0.151***
	Peru	0.050**	0.082***	0.087***	0.058***	0.081**	0.064***	0.056***	0.034***	-0.131***	-0.146***	-0.142***	-0.093***
	Argentina	-0.101***	-0.122***	-0.126***	-0.111***	n.s.	0.060***	0.056***	0.037***	0.092***	0.063***	0.071***	0.074***
	Bolivia	n.s.	n.a.	0.047*	-0.056***	n.s.	n.a.	n.s.	n.s.	n.s.	n.a.	n.s.	n.s.
more than 65	Chile	-0.089***	-0.081***	-0.072***	-0.077***	-0.026*	-0.027**	0.022**	-0.027**	0.115***	0.108***	0.050***	0.104***
years old	Colombia	n.s.	-0.040**	-0.043***	-0.040***	n.s.	-0.038**	-0.063***	-0.054***	n.s.	0.077***	0.106***	0.094***
N	Mexico	n.s.	n.s.	n.s.	-0.026***	n.s.	n.s.	-0.065***	-0.033**	n.s.	0.043**	0.053***	0.059***
	Peru	-0.032*	-0.041***	-0.038***	-0.031***	-0.125***	-0.070***	-0.095***	-0.062***	0.157***	0.111***	0.133***	0.093***

Annex 3.B: Age marginal effects of MNP - Model 1 (2000-2010)

EDUCATION	EDUCATION		Po	or			Middle	class		Affluent			
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010
Primary educat	ion	Ref.	Ref.	Ref.	Ref.	Ref.							
	Argentina	-0.130***	-0.073***	-0.101***	-0.094***	-0.142***	-0.119***	-0.099***	-0.062***	0.272***	0.192***	0.199***	0.156***
	Bolivia	-0.104***	n.a.	-0.084***	-0.090***	n.s.	n.a.	-0.087***	n.s.	0.125***	n.a.	0.171***	0.129***
Secondary	Chile	-0.143***	-0.119***	-0.117***	-0.096***	-0.130***	-0.112***	-0.102***	-0.120***	0.273***	0.230***	0.219***	0.216***
education	Colombia	-0.159***	-0.191***	-0.162***	-0.128***	-0.139***	-0.077***	-0.078***	-0.075***	0.298***	0.268***	0.240***	0.203***
	Mexico	-0.120***	-0.113***	-0.116***	-0.106***	-0.111***	-0.158***	-0.135***	-0.119***	0.230***	0.271***	0.251***	0.225***
	Peru	-0.127***	-0.117***	-0.118***	-0.086***	-0.100***	-0.091***	-0.062***	-0.076***	0.226***	0.208***	0.180***	0.162***
	Argentina	-0.169***	-0.120***	-0.131***	-0.196***	-0.355***	-0.257***	-0.286***	-0.243***	0.524***	0.377***	0.417***	0.440***
Technical	Bolivia	-0.202***	n.a.	-0.225***	-0.144***	-0.244***	n.a.	-0.271***	-0.198***	0.447***	n.a.	0.496***	0.342***
aducation or	Chile	-0.205***	-0.184***	-0.167***	-0.149***	-0.466***	-0.464***	-0.438***	-0.466***	0.671***	0.648***	0.606***	0.615***
University	Colombia	-0.200***	-0.299***	-0.225***	-0.181***	-0.416***	-0.171***	-0.346***	-0.377***	0.616***	0.469***	0.570***	0.559***
University	Mexico	-0.204***	-0.180***	-0.184***	-0.169***	-0.344***	-0.385***	-0.372***	-0.357***	0.547***	0.565***	0.556***	0.526***
	Peru	-0.179***	-0.171***	-0.170***	-0.155***	-0.375***	-0.300***	-0.290***	-0.271***	0.554***	0.471***	0.460***	0.427***

Annex 3.C: Education marginal effects of MNP - Model 1 (2000-2010)

MATRIMONI	MATRIMONIAL STATUS		Po	or			Middle	e class		Affluent			
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010
Single		Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
	Argentina	n.s.	n.s.	n.s.	-0.041***	n.s.	n.s.	0.033***	0.024**	-0.038**	n.s.	-0.022*	0.017*
	Bolivia	n.s.	n.a.	n.s.	n.s.	n.s.	n.a.	0.075**	0.064**	n.s.	n.a.	n.s.	-0.095***
Couplo	Chile	n.s.	n.s.	n.s.	-0.029***	n.s.	0.019*	n.s.	n.s.	n.s.	-0.020*	n.s.	0.018*
Couple	Colombia	n.s.	n.s.	n.s.	n.s.	n.s.	0.044***	0.041**	0.034**	n.s.	-0.037**	-0.061***	n.s.
	Mexico	0.031**	0.023*	0.033***	0.023***	n.s.	n.s.	0.030*	0.046***	-0.052*	n.s.	-0.064***	-0.069***
	Peru	0.050**	0.045***	0.035***	n.s.	n.s.	n.s.	n.s.	0.033**	-0.081*	-0.048**	-0.043***	-0.047***

Annex 3.D: Matrimonial status marginal effects of MNP - Model 1 (2000-2010)

ORIGIN			Po	oor		Middle class				Affluent			
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010
Majority group		Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
	Argentina	n.a.	n.a.	n.a.	n.a.	n.a.	n.a.	n.a.	n.a.	n.a.	n.a.	n.a.	n.a.
	Bolivia	0.140***	n.a.	0.090***	0.076***	0.039*	n.a.	n.s.	0.040*	-0.179***	n.a.	-0.101***	-0.116***
Ethnic group	Chile	0.081***	0.080***	0.029***	0.048***	n.s.	n.s.	n.s.	n.s.	-0.062***	-0.074***	-0.041***	-0.033**
Ethnic group	Colombia	n.a.	0.084***	n.a.	0.041***	n.a.	n.s.	n.a.	-0.029*	n.a.	-0.065***	n.a.	n.s.
	Mexico	n.a.	n.a.	n.a.	0.084***	n.a.	n.a.	n.a.	n.s.	n.a.	n.a.	n.a.	-0.071***
	Peru	n.a.	n.a.	0.121***	0.068***	n.a.	n.a.	n.s.	n.s.	n.a.	n.a.	-0.107***	-0.079***

Note: n.a. not available; n.s. not significant

Annex 3.E: Ethnicity marginal effects of MNP - Model 1 (2000-2010)

		Poor				Middle class				Affluent			
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010
Region metropol Capital	itana /	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
	Argentina	-0.047***	-0.017*	n.s.	-0.053***	-0.056***	-0.045***	-0.043***	-0.021***	0.104***	0.062***	0.035***	0.074***
	Bolivia	n.s.	n.a.	0.094***	0.103***	0.086***	n.a.	n.s.	n.s.	-0.083***	n.a.	-0.123***	-0.135***
Other provinces	Chile	-0.053***	-0.061***	-0.041***	-0.031***	n.s.	-0.024***	-0.035***	-0.039***	0.052***	0.085***	0.076***	0.070***
/ regions	Colombia	-0.082***	-0.128***	-0.152***	-0.097***	-0.054**	-0.038***	n.s.	-0.046***	0.137***	0.166***	0.181***	0.143***
	Mexico	-0.072***	-0.064***	-0.051***	-0.045***	n.s.	0.039**	0.065***	n.s.	0.083***	0.025*	n.s.	0.034***
	Peru	-0.157***	-0.112***	-0.115***	-0.113***	-0.103***	-0.188***	-0.140***	-0.081***	0.260***	0.300***	0.255***	0.193***

<u>Annex 3.F:</u> Geographic localization marginal effects of MNP - Model 1 (2000-2010)

			Po	or			Middl	e class		Affluent			
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010
Active occupied, services	other	Ref.											
	Argentina	0.122**	0.175***	n.s.	0.122***	-0.205***	-0.129***	-0.077**	-0.096**	n.s.	n.s.	n.s.	n.s.
	Bolivia	0.506***	n.a.	0.517***	0.365***	-0.182***	n.a.	-0.208***	-0.121***	-0.324***	n.a.	-0.308***	-0.244***
Active occupied,	Chile	0.115***	0.106***	0.099***	0.068***	n.s.	n.s.	n.s.	0.048***	-0.096***	-0.104***	-0.103***	-0.117***
agriculture	Colombia	0.327***	0.172***	0.247***	0.203***	-0.082***	n.s.	-0.033*	-0.050***	-0.246***	-0.183***	-0.214***	-0.154***
	Mexico	0.350***	0.242***	0.260***	0.233***	-0.140***	-0.120***	-0.139***	-0.102***	-0.210***	-0.123***	-0.121***	-0.131***
	Peru	0.242***	0.278***	0.294***	0.288***	-0.073*	-0.044**	-0.098***	-0.119***	-0.169***	-0.235***	-0.196***	-0.169***
	Argentina	-0.046***	n.s.	-0.043***	n.s.	0.051**	n.s.	n.s.	n.s.	n.s.	n.s.	0.024*	n.s.
	Bolivia	0.132***	n.a.	n.s.	n.s.	n.s.	n.a.	n.s.	n.s.	-0.076**	n.a.	n.s.	n.s.
Active occupied,	Chile	-0.017*	0.024**	n.s.	n.s.	0.079***	0.038***	0.028**	0.067***	-0.062***	-0.063***	-0.040***	-0.080***
industry	Colombia	n.s.											
-	Mexico	n.s.	n.s.	n.s.	-0.020**	-0.083***	n.s.	n.s.	0.059***	0.084***	n.s.	n.s.	-0.038***
	Peru	n.s.	0.051**	n.s.	0.058***	n.s.	-0.061**	-0.036*	-0.036*	n.s.	n.s.	n.s.	n.s.
	Argentina	n.s.	n.s.	n.s.	0.085***	n.s.	-0.107***						
	Bolivia	n.s.	n.a.	n.s.	n.s.	-0.081**	n.a.	-0.069*	-0.072**	0.088**	n.a.	n.s.	0.059*
Active occupied,	Chile	n.s.	0.030***	n.s.	n.s.	n.s.	n.s.	n.s.	0.048***	n.s.	-0.032**	n.s.	-0.050***
trade	Colombia	n.s.	0.066***	n.s.	-0.055***	n.s.	n.s.						
	Mexico	n.s.	n.s.	n.s.	-0.022**	n.s.	n.s.	n.s.	0.040***	n.s.	n.s.	n.s.	n.s.
	Peru	n.s.	n.s.	n.s.	0.031**	n.s.							
	Argentina	-0.033**	n.s.	-0.038***	0.022**	n.s.	-0.047***	n.s.	-0.029**	0.038**	0.043***	0.056***	n.s.
Active ecoupied	Bolivia	n.s.	n.a.	0.101*	n.s.	n.s.	n.a.	-0.114***	n.s.	n.s.	n.a.	n.s.	n.s.
Active occupied,	Chile	n.s.	0.034***	0.034***	0.015*	n.s.	n.s.	n.s.	0.028**	-0.025*	-0.033***	-0.042***	-0.043***
	Colombia	n.s.	-0.053**	-0.049**	n.s.	n.s.	-0.099***	-0.090***	-0.106***	n.s.	0.153***	0.139***	0.093***
Social services	Mexico	-0.094***	-0.082***	-0.060***	-0.033***	n.s.	n.s.	-0.054***	n.s.	0.066**	0.106***	0.115***	0.038***
	Peru	n.s.	0.074**	0.042*	0.046**	n.s.	-0.048*	-0.048**	n.s.	n.s.	n.s.	n.s.	-0.027*
	Argentina	0.330***	0.286***	0.301***	0.277***	-0.098***	-0.048**	-0.048**	-0.149***	-0.233***	-0.239***	-0.253***	-0.128***
	Bolivia	0.503***	n.a.	0.521***	0.475***	-0.193***	n.a.	-0.208***	-0.189***	-0.309***	n.a.	-0.312***	-0.286***
Active	Chile	0.460***	0.502***	0.481***	0.450***	-0.193***	-0.226***	-0.233***	-0.194***	-0.266***	-0.276***	-0.248***	-0.256***
unemployed	Colombia	0.510***	0.389***	0.354***	0.305***	-0.240***	-0.158***	-0.193***	-0.184***	-0.270***	-0.231***	-0.161***	-0.121***
	Mexico	n.s.	0.168***	n.s.	0.246***	0.144*	n.s.	n.s.	-0.089***	-0.126**	-0.140***	n.s.	-0.158***
	Peru	0.242***	0.348***	0.262***	0.302***	-0.145**	-0.167***	-0.126***	-0.145***	-0.097*	-0.181***	-0.136***	-0.157***
	Argentina	0.095***	0.074***	0.101***	0.176***	0.031*	n.s.	0.067***	n.s.	-0.126***	-0.074***	-0.168***	-0.189***
	Bolivia	0.444***	n.a.	0.283***	0.330***	-0.219***	n.a.	-0.146***	-0.142***	-0.225***	n.a.	-0.137***	-0.188***
Incotivo	Chile	0.109***	0.196***	0.212***	0.220***	n.s.	-0.040***	-0.038***	n.s.	-0.110***	-0.157***	-0.174***	-0.213***
mactive	Colombia	0.280***	0.434***	0.311***	0.272***	-0.135***	-0.171***	-0.138***	-0.111***	-0.144***	-0.263***	-0.173***	-0.161***
N	Mexico	0.057***	0.047***	0.039**	0.045***	n.s.	-0.050**	-0.051***	-0.058***	n.s.	n.s.	n.s.	n.s.
F	Peru	n.s.	0.141***	0.221***	0.138***	n.s.	-0.153***	-0.170***	-0.095***	n.s.	n.s.	-0.050**	-0.043**

Annex 3.G: Sector of activity marginal effects of MNP	- Model 1 (2000-2010)
---	-----------------------

			Po	or			Middle	class		Affluent					
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010		
Not self employed		Ref.													
	Argentina	0.138***	0.133***	0.098***	0.126***	-0.049***	n.s.	0.026**	n.s.	-0.090***	-0.130***	-0.124***	-0.110***		
	Bolivia	0.244***	n.a.	0.185***	0.193***	-0.080***	n.a.	-0.065***	-0.047**	-0.164***	n.a.	-0.120***	-0.146***		
Solfomployed	Chile	-0.036***	-0.060***	-0.047***	-0.040***	-0.045***	-0.093***	-0.109***	-0.147***	0.081***	0.153***	0.156***	0.187***		
Sellemployed	Colombia	0.152***	0.596***	0.220***	0.170***	-0.057***	-0.160***	-0.073***	n.s.	-0.094***	-0.436***	-0.146***	-0.173***		
	Mexico	n.s.	0.064***	0.055***	0.075***	0.114***	n.s.	0.027*	-0.068***	-0.125***	-0.054***	-0.082***	n.s.		
	Peru	0.158***	0.165***	0.145***	0.144***	-0.070**	n.s.	n.s.	-0.028**	-0.087***	-0.163***	-0.147***	-0.116***		

<u>Annex 3.H:</u> Self-employment marginal effects of MNP - Model 1 (2000-2010)

Annex 4: Other household members' activity status, model 1

As expected, in all countries, although in different proportions, the number of other occupied members in the household displays a positive effect on income classes belonging while the number of unemployed and inactive members in the household displays a negative one.

Indeed, in 2009-2010, each additional occupied member in the household decreases the likelihood of being poor and increases the likelihood of being affluent (annex 4.A). The impact on the middle sector is not significant in 2009-2010. On previous years, two patterns emerge: In Chile (and Colombia only in 2003) the effect is positive on middle class, additional occupied household members being a middle class determinant as well as an affluent one, and thus prevent from falling into poverty. In Bolivia, Peru (and Argentina only in 2006), the effect is negative on middle class, additional occupied household members being a determinant of the affluent class only.

High impacts of additional occupied members are noticeable in Chile and Colombia while low impacts are observed in Peru and Mexico.

Considering time evolution of these effects, we can identify 3 patterns: Occupied members in the household; we observe an increasing impact along the years in Chile and Mexico. On the contrary, we observe a decreasing impact along the years in Argentina and Colombia. Regarding Peru and Bolivia, there is an increasing impact along the years regarding the poor, while it is the contrary as far as the middle sector and the affluent are concerned.

			Po	or			Middle	e class		Affluent					
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010		
	Argentina	-0.090***	-0.047***	-0.040***	-0.065***	n.s.	n.s.	-0.012**	n.s.	0.101***	0.041***	0.052***	0.066***		
Number of other	Bolivia	-0.044***	n.a.	-0.041***	-0.049***	-0.030**	n.a.	-0.024**	n.s.	0.074***	n.a.	0.065***	0.061***		
occupied	Chile	-0.112***	-0.123***	-0.124***	-0.136***	0.016***	0.016***	0.023***	n.s.	0.095***	0.107***	0.101***	0.130***		
household	Colombia	-0.129***	-0.106***	-0.102***	-0.092***	n.s.	0.019***	n.s.	n.s.	0.121***	0.087***	0.100***	0.093***		
members	Mexico	-0.038***	-0.031***	-0.015***	-0.044***	n.s.	n.s.	n.s.	n.s.	0.034***	0.028***	0.013***	0.042***		
	Peru	-0.029***	-0.020***	-0.030***	-0.036***	-0.021*	-0.014**	n.s.	n.s.	0.050***	0.035***	0.034***	0.041***		

Annex 4.A: Other occupied members marginal effects of MNP Model 1 (2000-2010)

Annex 4.B shows that each additional unemployed member in the household increases the likelihood of being poor or middle sector, and decreases the likelihood of being affluent. The highest impacts are displayed in Argentina and Chile while the smallest one is observed in Peru and Bolivia (except for middle sectors). We observe an increasing impact along the years in Argentina and in Peru regarding the affluent while the impact decreases along the years in the other countries and in Peru regarding the poor and the middle sector.

										-						
			Po	or			Middle	e class		Affluent						
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010			
	Argentina	0.069***	0.111***	0.109***	0.094***	0.055***	0.051***	0.057***	0.060***	-0.124***	-0.163***	-0.167***	-0.154***			
Number of other	Bolivia	n.s.	n.a.	n.s.	n.s.	0.100***	n.a.	n.s.	0.065**	-0.082***	n.a.	n.s.	-0.070**			
unemployed	Chile	0.102***	0.105***	0.102***	0.081***	0.062***	0.068***	0.052***	0.050***	-0.164***	-0.172***	-0.154***	-0.131***			
household	Colombia	0.067***	-0.027*	n.s.	0.057***	0.045**	0.045***	0.068***	0.035*	-0.112***	n.s.	-0.084***	-0.092***			
members	Mexico	0.048**	0.036**	0.066***	0.059***	n.s.	0.058**	0.087***	0.035**	-0.109***	-0.095***	-0.153***	-0.093***			
	Peru	-0.044***	n.s.	n.s.	0.023***	0.087***	0.082***	0.069***	0.051***	-0.043*	-0.081***	-0.078***	-0.074***			

Annex 4.B: Other unemployed members marginal effects of MNP Model 1 (2000-2010)

Each additional inactive member in the household increases the likelihood of being poor or middle sectors and decreases the likelihood of being affluent (annex 4.C). The highest impact is displayed in Argentina while the slightest ones are observed in Peru and Bolivia. The impact increases along the years in Argentina (all social classes), Bolivia, Colombia and Mexico (middle sectors and affluent only regarding those last three countries). On the contrary the impact decreases in Chile (except regarding the middle sectors) and in Peru.

			Po	or			Middle	eclass		Affluent						
		2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010	2000	2003/2004	2005/2006	2009/2010			
	Argentina	0.052***	0.051***	0.064***	0.083***	0.041***	0.036***	0.052***	0.068***	-0.093***	-0.086***	-0.116***	-0.150***			
Number of other	Bolivia	n.s.	n.a.	n.s.	0.024**	0.037***	n.a.	0.067***	0.046***	-0.035***	n.a.	-0.050***	-0.070***			
inactive	Chile	0.056***	0.057***	0.054***	0.048***	0.039***	0.035***	0.038***	0.046***	-0.096***	-0.092***	-0.092***	-0.094***			
household	Colombia	0.051***	n.s.	0.057***	0.029***	0.026***	0.054***	0.052***	0.086***	-0.077***	-0.056***	-0.109***	-0.115***			
members	Mexico	0.046***	0.042***	0.044***	0.045***	0.024***	0.042***	0.037***	0.045***	-0.069***	-0.084***	-0.081***	-0.090***			
	Peru	-0.057***	-0.031***	n.s.	0.024***	n.s.	0.044***	0.021**	0.023***	0.050***	n.s.	-0.021**	-0.046***			

<u>Annex 4.C:</u> Other inactive members marginal effects of MNP Model 1 (2000-2010)

In conclusion, active occupied members increase the likelihood of affluency and decrease the likelihood of being poor. Unemployed and inactive members are determinants of poor and middle class and not of the affluent class.

	ARGENTINA				BOLIVIA		CHILE			COLOMBIA			MEXICO			PERI		
		2010			2009			2009			2010			2010			2010	
	Poor	Middle	Affluent	Poor	Middle	Affluent	Poor	Middle	Affluent	Poor	Middle	Affluent	Poor	Middle	Affluent	Poor	Middle	Affluent
VARIABLES	1 001	sector	Andent	1 001	sector	Andent	1 001	sector	Andent	1 001	sector	Andent	1 001	sector	Andent	1.001	sector	Andent
						/									/			
Male	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
Female	0.027	0.088***	-0.115***	0.207**	0.109	-0.317***	0.034	0.087**	-0.121***	0.154***	0.023	-0.177***	0.107**	-0.115**	0.008	0.094**	0.064	-0.158***
	(0.021)	(0.028)	(0.025)	(0.084)	(0.086)	(0.049)	(0.025)	(0.037)	(0.037)	(0.036)	(0.044)	(0.038)	(0.049)	(0.051)	(0.042)	(0.037)	(0.041)	(0.034)
up to 30 years old	0.128	0.051	-0.179***	0.004	0.120***	-0.124***	0.032	0.061	-0.093***	0.051	0.107***	-0.157	0.161	0.028"	-0.189***	0.065	0.060	-0.125
21 10 years ald	(0.015)	(0.016)	(0.010)	(0.027)	(0.033)	(0.028)	(0.012)	(0.016)	(0.015)	(0.018)	(0.021)	(0.018)	(0.016)	(0.016)	(0.010)	(0.016)	(0.018)	(0.015)
31-40 years old	0.100	0.003	-0.103	0.008	0.093	-0.101	0.032	0.036	-0.068	0.035	0.089	-0.123	0.100	0.044	-0.145	0.064	0.032	-0.096
11.05 wears ald	(0.013)	(0.014)	(0.011)	(0.024)	(0.030)	(0.026)	(0.008)	(0.012)	(0.011)	(0.015)	(0.018)	(0.017)	(0.011)	(0.013)	(0.010)	(0.011)	(0.014)	(0.012)
41-65 years old	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
more than 65 years old	-0.114	0.037	0.077	-0.080	0.049	0.032	-0.074	-0.030	0.103	-0.026	-0.052	0.078	-0.033	0.003	0.031	-0.044	-0.054	0.098
Female*0.20 years ald	(0.009)	(0.018)	(0.019)	(0.022)	(0.044)	(0.044)	(0.007)	(0.017)	(0.019)	(0.015)	(0.025)	(0.027)	(0.010)	(0.018)	(0.018)	(0.009)	(0.018)	(0.019)
Female 0-30 years old	0.001	-0.125	0.124	0.068	-0.018	-0.050	0.000	-0.059	-0.008	0.015	0.017	-0.033	-0.005	0.001	0.004	0.040	-0.070	0.030
Female*24, 44 years ald	(0.020)	(0.028)	(0.029)	(0.062)	(0.070)	(0.069)	(0.024)	(0.033)	(0.036)	(0.029)	(0.040)	(0.039)	(0.023)	(0.035)	(0.035)	(0.037)	(0.045)	(0.045)
Female 31-41 years old	0.001	-0.036	0.035	-0.032	-0.020	0.052	0.023	(0.021	-0.044 ^{**}	-0.024	0.075	-0.050	-0.012	0.055	-0.043	-0.016	0.003	0.013
Female*Mere CF vegee ald	(0.017)	(0.025)	(0.025)	(0.045)	(0.005)	(0.000)	(0.014)	(0.022)	(0.023)	(0.021)	(0.033)	(0.032)	(0.016)	(0.027)	(0.020)	(0.020)	(0.032)	(0.033)
Female More 65 years old	0.009	-0.017	0.008	0.080	-0.170	0.090	-0.015	-0.009	0.024	-0.042	-0.011	0.054	0.015	-0.088	0.073	0.043	-0.033	-0.010
Brimony advantion	(0.020)	(0.026) Ref	(0.027) Ref	(0.064)	(0.070) Rof	(0.080) Ref	(0.012) Rof	(0.026) Ref	(0.031) Ref	(0.021) Ref	(0.041) Rof	(0.043) Rof	(0.021) Ref	(0.031) Ref	(0.032) Rof	(0.022) Rof	(0.031) Ref	(0.032) Ref
Secondary education	0.005***	0.072***	0 167***	0.002***	0.024	0 127***	0.008***	0 121***	0.219***	0 120***	0.062***	0 192***	-0.101***	0 105***	0.207***	0.086***	0.075***	0 161***
Secondary education	-0.095	(0.073	(0.013)	-0.093	(0.029)	(0.020)	(0.005)	(0.010)	(0.011)	-0.120	-0.003	(0.018)	(0.006)	(0.014)	(0.014)	(0.007)	(0.015)	(0.015)
Technical education or University	-0.187***	-0.261***	0.448***	-0.142***	-0 166***	0.308***	-0.146***	-0.469***	0.614***	-0 172***	-0.373***	0.546***	-0.165***	-0.338***	0.503***	-0.156***	-0.282***	0 /37***
reclinical education of Oniversity	-0.107	-0.201	(0.012)	-0.142	-0.100	(0.033)	-0.140	-0.409	(0.014)	-0.172	-0.373	(0.020)	-0.105	-0.330	(0.012)	(0.007)	-0.202	(0.017)
Female*Secondary education	0.007	0.026	-0 033*	0.054	-0.041	-0.013	0.000	0.003	-0.003	-0.026	-0.037	0.020)	-0.019	-0.038	0.056**	-0.003	-0.017	0.020
I emale becondary education	(0.014)	(0.020)	(0.020)	(0.053)	(0.063)	(0.058)	(0.010)	(0.000	(0.021)	(0.020	(0.031)	(0.034)	(0.015)	(0.026)	(0.026)	(0.018)	(0.029)	(0.020)
Female*Technical education or University	-0.030*	0.058***	-0.028	-0.034	-0.167**	0.200***	-0.037*	0.029	0.009	-0.047*	-0.004	0.052	-0.018	-0.071**	0.090***	0.003	0.035	-0.038
I childle rechilical education of childensity	(0.016)	(0.022)	(0.020)	(0.059)	(0.069)	(0.076)	(0.019)	(0.028)	(0.028)	(0.028)	(0.043)	(0.041)	(0.023)	(0.031)	(0.030)	(0.032)	(0.037)	(0.032)
Single	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref	Ref
Couple	-0.046***	0.049***	-0.002	0.073***	0.124***	-0.197***	-0.023**	0.043***	-0.021	-0.012	0.058***	-0.046**	0.053***	0.037**	-0.090***	0.034***	0.033**	-0.067***
	(0.010)	(0.013)	(0.013)	(0.026)	(0.036)	(0.037)	(0.009)	(0.014)	(0.015)	(0.016)	(0.021)	(0.022)	(0.011)	(0.017)	(0.016)	(0.010)	(0.017)	(0.017)
Female*Couple	0.013	-0.060***	0.047**	-0.107***	-0.227***	0.334***	-0.016	-0.085***	0.101***	-0.013	-0.073**	0.086**	-0.066***	0.001	0.065**	-0.068***	-0.017	0.086**
	(0.016)	(0.022)	(0.022)	(0.026)	(0.056)	(0.062)	(0.011)	(0.021)	(0.024)	(0.022)	(0.033)	(0.035)	(0.012)	(0.029)	(0.029)	(0.015)	(0.034)	(0.036)
Majority group	, í			Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
Ethnic group				0.100***	0.029	-0.129***	0.036***	-0.004	-0.032*	0.040***	-0.024	-0.016	0.091***	-0.016	-0.075***	0.070***	0.016	-0.086***
	N.A.	N.A.	N.A.	(0.019)	(0.025)	(0.023)	(0.009)	(0.017)	(0.018)	(0.015)	(0.020)	(0.021)	(0.009)	(0.012)	(0.011)	(0.008)	(0.011)	(0.011)
Female*Ethnic group				-0.073**	0.025	0.048	0.038**	-0.027	-0.011	0.003	-0.015	0.012	-0.019	0.006	0.013	-0.009	-0.023	0.032
				(0.030)	(0.049)	(0.048)	(0.017)	(0.031)	(0.033)	(0.023)	(0.037)	(0.039)	(0.013)	(0.023)	(0.023)	(0.014)	(0.024)	(0.025)
Other provinces/regions	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.
Capital Buenos Aires	-0.050***	-0.019**	0.070***	0.098***	0.052**	-0.151***	-0.031***	-0.030***	0.061***	-0.096***	-0.052**	0.148***	-0.045***	0.012	0.033***	-0.107***	-0.090***	0.198***
	(0.007)	(0.009)	(0.009)	(0.023)	(0.026)	(0.023)	(0.006)	(0.010)	(0.010)	(0.015)	(0.022)	(0.022)	(0.007)	(0.011)	(0.010)	(0.010)	(0.014)	(0.014)
Female*Capital	-0.005	-0.011	0.016	0.012	-0.093*	0.081	-0.001	-0.024	0.026	-0.000	0.015	-0.014	-0.007	0.003	0.004	-0.016	0.041	-0.024
	(0.011)	(0.016)	(0.015)	(0.039)	(0.052)	(0.056)	(0.010)	(0.018)	(0.019)	(0.032)	(0.037)	(0.033)	(0.015)	(0.021)	(0.019)	(0.024)	(0.028)	(0.024)

Annex 5A: Multinomial Probit estimation of variable poor/middle sector/affluent with female specific effects (Part 1)

	ARGENTINA			BOLIVIA			CHILE			COLOMBIA			MEXICO			PERU			
		2010			2009			2009			2010			2010			2010		
	Poor	Middle	Affluent																
VARIABLES	1 001	sector	7 maoni	1 001	sector	7 undern	1 001	sector	/ indoni	1 001	sector	7 indent	1 001	sector	Amaoni	1 001	sector	Amaoni	
No we have a fact the second start to second a start of the second starts	0.050***	0.007	0.005+++	0.000***	0.004	0.040***	0.400***	0.005	0.405***	0.000***	0.000	0.004***	0.040***	0.000	0.044***	0.000***	0.000	0.000***	
Number of other occupied household members	-0.059***	-0.007	0.065	-0.039***	-0.004	0.043	-0.120***	-0.005	0.125	-0.086***	-0.006	0.091	-0.043***	0.002	0.041	-0.029***	-0.003	0.032	
Fomolo*Number of other occupied household members	(0.005)	(0.006)	(0.006)	(0.010)	(0.012)	(0.011)	(0.005)	(0.006)	(0.006)	(0.007)	(0.008)	(0.009)	(0.004)	(0.006)	(0.005)	(0.003)	(0.005)	(0.005)	
Female Number of other occupied household members	(0.008)	(0.010)	-0.001	(0.022)	-0.035	(0.024)	(0.043	(0.033	(0.010	-0.018	(0.014	(0.003	-0.005	-0.000	(0.005	(0.007)	-0.009	(0.040	
Number of other unemployed household members	0.084***	0.077***	-0.161***	0.014	0.058*	-0.072**	0.077***	0.057***	-0 134***	0.071***	0.020	-0.090***	0.057***	0.027	-0.084***	0.022***	0.053***	-0.075***	
	(0.011)	(0.019)	(0.020)	(0.025)	(0.033)	(0.033)	(0.006)	(0.013)	(0.014)	(0.015)	(0.026)	(0.026)	(0.012)	(0.019)	(0.018)	(0.007)	(0.010)	(0.010)	
Female*Number of other unemployed household members	0.031*	-0.058*	0.027	-0.030	0.015	0.015	0.014	-0.024	0.009	-0.031	0.035	-0.005	0.009	0.035	-0.044	0.006	-0.000	-0.006	
	(0.018)	(0.033)	(0.037)	(0.066)	(0.089)	(0.087)	(0.010)	(0.022)	(0.024)	(0.024)	(0.039)	(0.041)	(0.021)	(0.034)	(0.035)	(0.017)	(0.024)	(0.025)	
Number of other inactive household members	0.083***	0.070***	-0.153***	0.028***	0.041***	-0.069***	0.050***	0.047***	-0.096***	0.031***	0.088***	-0.119***	0.043***	0.045***	-0.088***	0.014***	0.027***	-0.041***	
	(0.004)	(0.006)	(0.007)	(0.011)	(0.014)	(0.013)	(0.003)	(0.005)	(0.006)	(0.006)	(0.009)	(0.010)	(0.004)	(0.006)	(0.006)	(0.005)	(0.008)	(0.007)	
Female*Number of other inactive household members	0.000	-0.020*	0.019	-0.010	0.010	0.000	-0.004	-0.016	0.019*	-0.007	-0.014	0.021	0.002	0.001	-0.002	0.043***	-0.012	-0.030*	
	(0.007)	(0.012)	(0.013)	(0.024)	(0.033)	(0.032)	(0.005)	(0.010)	(0.012)	(0.011)	(0.018)	(0.019)	(0.008)	(0.014)	(0.014)	(0.011)	(0.016)	(0.017)	
Active occupied, agriculture	0.126***	-0.095^^	-0.031	0.399***	-0.146***	-0.254***	0.069***	0.045	-0.114***	0.224***	-0.060***	-0.164***	0.235***	-0.109***	-0.126***	0.288***	-0.116^^^	-0.172***	
Active accupied industry	(0.035)	(0.041)	(0.039)	(0.034)	(0.034)	(0.024)	(0.010)	(0.012)	(0.010)	(0.020)	(0.020)	(0.017)	(0.017)	(0.018)	(0.015)	(0.017)	(0.017)	(0.013)	
Active occupied, industry	(0.013)	(0.016)	(0.003	(0.037)	(0.038)	-0.043	(0.010)	(0.015)	-0.008	(0.003	(0.026)	(0.025)	-0.022	(0.031	-0.029	(0.033	(0.023)	(0.020	
Active occupied trade	0.082***	0.025	-0 107***	0.006	-0.050	0.045	0.012	0.033**	-0.045***	0.001	-0.018	0.017	-0.015	0.026	-0.011	0.050***	-0.018	-0.032*	
Nonve boodpied, hade	(0.014)	(0.016)	(0.013)	(0.037)	(0.042)	(0.040)	(0.011)	(0.015)	(0.014)	(0.018)	(0.023)	(0.023)	(0.011)	(0.017)	(0.016)	(0.019)	(0.020)	(0.018)	
Active occupied, public and social services	-0.045***	-0.006	0.052***	-0.033	-0.001	0.033	-0.001	0.028**	-0.027**	-0.014	-0.124***	0.138***	-0.038***	-0.016	0.054***	0.052**	-0.005	-0.047***	
	(0.013)	(0.017)	(0.016)	(0.042)	(0.043)	(0.038)	(0.010)	(0.014)	(0.013)	(0.030)	(0.032)	(0.033)	(0.010)	(0.015)	(0.014)	(0.022)	(0.022)	(0.017)	
Active occupied, other services	Ref.																		
Active unemployed	0.296***	-0.179***	-0.117***	0.550***	-0.265***	-0.285***	0.452***	-0.197***	-0.255***	0.329***	-0.193***	-0.136***	0.263***	-0.114***	-0.149***	0.383***	-0.209***	-0.173***	
	(0.033)	(0.030)	(0.025)	(0.080)	(0.077)	(0.018)	(0.025)	(0.024)	(0.011)	(0.037)	(0.032)	(0.025)	(0.032)	(0.030)	(0.021)	(0.037)	(0.033)	(0.017)	
Inactive	0.166***	0.037**	-0.203***	0.478***	-0.255***	-0.224***	0.208***	0.020	-0.228***	0.318***	-0.138***	-0.179***	0.053***	-0.102***	0.049**	0.189***	-0.117***	-0.072***	
	(0.016)	(0.017)	(0.014)	(0.063)	(0.054)	(0.032)	(0.014)	(0.016)	(0.016)	(0.035)	(0.033)	(0.025)	(0.017)	(0.021)	(0.019)	(0.032)	(0.031)	(0.024)	
Female*Active occupied, agriculture	-0.065	0.005	0.060	-0.086**	0.087	-0.001	0.015	0.051	-0.066	-0.056**	0.011	0.045	0.007	0.117*	-0.124*	-0.009	-0.031	0.040	
Fomolo*Active accurring industry	(0.056)	(0.120)	(0.110)	(0.036)	(0.089)	(0.091)	(0.029)	(0.045)	(0.045)	(0.026)	(0.067)	(0.075)	(0.046)	(0.068)	(0.064)	(0.024)	(0.042)	(0.045)	
Female Active occupied, industry	-0.027	(0.005	(0.022	-0.049	-0.008	(0.002)	(0.062	-0.003	-0.079	(0.019	-0.030	(0.052)	-0.002	(0.049)	-0.113	(0.023	-0.041	(0.018	
Female*Active occupied trade	-0 109***	-0.110	0.218	-0.027	-0.052	0.086	-0.021	0.052	-0.031	-0.032	0.015	0.018	-0.032	0.141***	-0 110***	-0 044*	-0.004	0.047)	
	(0.021)	(0.151)	(0.159)	(0.059)	(0.084)	(0.086)	(0.021)	(0.037)	(0.036)	(0.028)	(0.045)	(0.045)	(0.032)	(0.042)	(0.032)	(0.023)	(0.040)	(0.041)	
Female*Active occupied, public and social services	0.148***	-0.065**	-0.083***	0.050	0.029	-0.080	0.049*	0.010	-0.059*	0.037	0.046	-0.083**	0.007	0.118***	-0.125***	-0.032	-0.081*	0.114**	
	(0.031)	(0.029)	(0.021)	(0.090)	(0.091)	(0.070)	(0.029)	(0.035)	(0.031)	(0.046)	(0.049)	(0.040)	(0.037)	(0.042)	(0.027)	(0.030)	(0.044)	(0.044)	
Female*Active unemployed	-0.022	0.092*	-0.070	-0.092*	0.273**	-0.180*	0.004	0.009	-0.013	-0.049	-0.028	0.077	-0.055	0.211***	-0.156***	-0.090***	0.042	0.048	
	(0.029)	(0.051)	(0.051)	(0.055)	(0.115)	(0.106)	(0.030)	(0.067)	(0.072)	(0.036)	(0.069)	(0.070)	(0.034)	(0.063)	(0.054)	(0.017)	(0.052)	(0.052)	
Female*Inactive	0.032	-0.049*	0.017	-0.153***	0.159*	-0.006	0.045	-0.041	-0.004	-0.052*	0.049	0.003	-0.014	0.171***	-0.156***	-0.076***	-0.017	0.093*	
	(0.022)	(0.028)	(0.027)	(0.020)	(0.086)	(0.086)	(0.028)	(0.038)	(0.040)	(0.027)	(0.047)	(0.047)	(0.034)	(0.040)	(0.027)	(0.020)	(0.048)	(0.049)	
No selfemployed	Ref.																		
Seirempioyea	0.120***	-0.011	-0.109^**	0.187***	-0.052*	-0.135^**	-0.051^**	-0.144^**	0.195***	0.178***	-0.012	-0.166^**	0.072***	-0.063^**	-0.009	0.149***	-0.019	-0.130***	
Female*Selfempleyed	(0.012)	(0.013)	(0.011)	(0.022)	(0.027)	(0.024)	(U.UUb)	(0.012)	(0.013)	(0.013)	(0.016)	(0.015)	(0.009)	(0.013)	(0.012)	(0.009)	(0.012)	(0.012)	
remaie Seliempioyed	(0.024	-0.007	-0.017	-0.007	(0.060)	-0.077	(0.032	-0.018	-0.035	-0.022	0.039°	-0.030	(0.019	-0.034	0.015	-0.024	-0.031	0.033°	
	(0.021)	(0.020)	(0.020)	(0.040)	(0.000)	(0.007)	(0.020)	(0.020)	(0.027)	(0.021)	(0.002)	(0.001)	(0.013)	(0.021)	(0.020)	(0.017)	(0.000)	(0.002)	
Observations	43,609	43,609	43,609	4,006	4,006	4,006	70,702	70,702	70,702	14,531	14,531	14,531	27,652	27,652	27,652	21,495	21,495	21,495	

Standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1

Annex 5B: Multinomial Probit estimation of variable poor/middle sector/affluent with female specific effects (Part 2)

	ARGENTINA				BOLIVIA			CHILE			COLOMBIA			MEXICO			PERU		
		2010			2009			2009			2010			2010			2010		
VARIABLES	Poor	Middle	Affluent	Poor	Middle	Affluent	Poor	Middle	Affluent	Poor	Middle	Affluent	Poor	Middle	Affluent	Poor	Middle	Affluent	
		sector	741140111		sector	741140111		sector			sector	711140111		sector	/		sector	/	
up to 30 years old	0 141***	0.023	-0 165***	0.011	0.082***	-0 094***	0.052***	0.064***	-0 115***	0 106***	0 102***	-0 208***	0 224***	-0.012	-0 212***	0 135***	0.021	-0 156***	
	(0.015)	(0.016)	(0.012)	(0.026)	(0.032)	(0.027)	(0.010)	(0.014)	(0.014)	(0.016)	(0.020)	(0.018)	(0.015)	(0.016)	(0.009)	(0.016)	(0.017)	(0.013)	
31-40 years old	0.116***	0.011	-0.127***	0.071***	0.059*	-0.130***	0.067***	0.054***	-0.121***	0.098***	0.085***	-0.183***	0.138***	0.034***	-0.172***	0.094***	0.050***	-0.145***	
	(0.012)	(0.014)	(0.012)	(0.027)	(0.031)	(0.025)	(0.008)	(0.012)	(0.011)	(0.015)	(0.019)	(0.018)	(0.012)	(0.013)	(0.010)	(0.012)	(0.014)	(0.012)	
41-65 years old	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	
more than 65 years old	-0.113***	-0.015	0.127***	-0.090***	0.028	0.062	-0.076***	-0.006	0.082***	-0.022	0.001	0.021	-0.054***	-0.007	0.060***	-0.015	-0.067***	0.082***	
Deimony a dua tian	(0.006)	(0.023)	(0.024)	(0.024)	(0.056)	(0.056)	(0.005)	(0.021)	(0.023)	(0.018)	(0.037)	(0.038)	(0.011)	(0.023)	(0.023)	(0.012)	(0.023)	(0.024)	
Primary education	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	
Secondary education	-0.062	-0.015	(0.016)	-0.049	(0.047	(0.002	-0.046	-0.044 (0.013)	(0.090	-0.069	-0.040	(0.022)	-0.074	-0.024	(0.099	-0.037	-0.030	(0.067	
Technical education or University	-0 134***	-0 128***	0.262***	-0.039	-0.085	0.123**	-0.076***	-0.260***	0.336***	-0 108***	-0.263***	0.370***	-0 122***	-0 170***	0.292***	-0.092***	-0 168***	0.259***	
	(0.009)	(0.019)	(0.019)	(0.053)	(0.056)	(0.053)	(0.008)	(0.020)	(0.021)	(0.015)	(0.031)	(0.032)	(0.009)	(0.019)	(0.019)	(0.012)	(0.023)	(0.023)	
Majority group	(01000)	(0.0.0)	(0.0.0)	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	
Ethnic group	N.A.	N.A.	N.A.	0.095***	0.010	-0.105***	0.023***	0.044***	-0.067***	0.026*	-0.006	-0.020	0.080***	-0.025**	-0.055***	0.051***	0.003	-0.053***	
				(0.020)	(0.027)	(0.024)	(0.008)	(0.017)	(0.018)	(0.015)	(0.022)	(0.023)	(0.009)	(0.013)	(0.012)	(0.008)	(0.012)	(0.012)	
Other provinces/regions	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	
Capital Buenos Aires	-0.059***	-0.027***	0.086***	0.083***	0.053*	-0.137***	-0.034***	-0.028***	0.062***	-0.085***	-0.070***	0.155***	-0.036***	0.025**	0.011	-0.089***	-0.081***	0.171***	
Active economical empirations	(0.006)	(0.010)	(0.010)	(0.024)	(0.028)	(0.023)	(0.005)	(0.011)	(0.011)	(0.015)	(0.025)	(0.025)	(0.008)	(0.011)	(0.010)	(0.011)	(0.016)	(0.015)	
Active occupied, agriculture	-0.100	0.080	(0.112)	(0.064)	0.015	-0.239	0.081*	(0.040)	-0.189***	0.045	0.011	-0.056	0.076	0.058	-0.134	(0.025)	-0.036	-0.171	
Active accupied industry	0.050*	0.003	-0.053*	(0.004)	-0.087	-0.109**	0.042)	0.040)	-0.083**	(0.033)	-0.071*	(0.040)	(0.032)	(0.052)	-0.106***	0.105***	-0.042	-0.062**	
reave becapied, industry	(0.028)	(0.033)	(0.027)	(0.079)	(0.069)	(0.045)	(0.043)	(0.044)	(0.036)	(0.031)	(0.042)	(0.041)	(0.020	(0.041)	(0.025)	(0.027)	(0.029)	(0.024)	
Active occupied, trade	-0.019	0.044	-0.025	0.095	0.010	-0.105***	0.072*	0.026	-0.099***	-0.020	-0.030	0.049	0.007	0.069*	-0.076***	0.022	-0.050**	0.029	
	(0.044)	(0.091)	(0.085)	(0.059)	(0.058)	(0.040)	(0.039)	(0.037)	(0.027)	(0.024)	(0.036)	(0.036)	(0.038)	(0.037)	(0.025)	(0.019)	(0.023)	(0.021)	
Active occupied, public and social services	0.023*	0.071***	-0.094***	-0.107**	0.020	0.087	0.120***	-0.000	-0.120***	0.020	-0.070*	0.050	-0.026	0.065*	-0.040	0.022	-0.009	-0.013	
	(0.013)	(0.017)	(0.014)	(0.042)	(0.060)	(0.056)	(0.041)	(0.037)	(0.026)	(0.031)	(0.036)	(0.036)	(0.033)	(0.034)	(0.026)	(0.026)	(0.028)	(0.023)	
Active occupied, other services	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	
Active unemployed	0.197***	0.042	-0.239***	0.242***	-0.074	-0.167***	0.363***	-0.076	-0.287***	0.198***	-0.077	-0.121***	0.066	0.112	-0.177***	0.164***	-0.020	-0.144***	
	(0.038)	(0.038)	(0.014)	(0.085)	(0.077)	(0.038)	(0.063)	(0.057)	(0.016)	(0.067)	(0.071)	(0.046)	(0.086)	(0.084)	(0.025)	(0.030)	(0.028)	(0.017)	
Inactive	0.126***	0.138***	-0.264^^^	0.173***	-0.001	-0.172^^^	0.190***	0.129***	-0.318^^^	0.128***	0.082***	-0.210^^^	0.085	0.096^^^	-0.182^^^	0.104^^^	-0.013	-0.091^^^	
No self employed	(0.011) Ref	(0.010) Ref	(0.014) Rof	(0.057) Ref	(0.055) Ref	(0.030) Ref	(0.020) Ref	(0.029) Rof	(0.020) Ref	(0.021) Ref	(0.030) Ref	(0.026) Ref	(0.031) Ref	(0.033) Ref	(0.028) Ref	(0.027) Ref	(0.020) Ref	(0.019) Ref	
Selfemployed	0.053***	0.053***	-0 106***	-0.034	0.027	0.007	0.054***	-0.096***	0.042*	0.058***	0.098***	-0 157***	0.036***	0.013	-0.050***	0.020*	0.075***	-0.095***	
Contemployed	(0.017)	(0.020)	(0.016)	(0.029)	(0.038)	(0.035)	(0.020)	(0.022)	(0.022)	(0.022)	(0.025)	(0.021)	(0.011)	(0.015)	(0.013)	(0.012)	(0.016)	(0.015)	
Husband Primary education	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	
Husband Secondary education	-0.056***	-0.039***	0.095***	-0.055**	-0.024	0.078**	-0.061***	-0.075***	0.136***	-0.074***	0.002	0.072***	-0.075***	-0.053***	0.128***	-0.043***	-0.028*	0.071***	
	(0.008)	(0.015)	(0.015)	(0.024)	(0.034)	(0.033)	(0.006)	(0.012)	(0.013)	(0.012)	(0.021)	(0.021)	(0.008)	(0.016)	(0.016)	(0.008)	(0.016)	(0.017)	
Husband Technical education or University	-0.110***	-0.186***	0.297***	-0.092**	-0.082*	0.173***	-0.116***	-0.314***	0.430***	-0.098***	-0.238***	0.335***	-0.129***	-0.216***	0.345***	-0.108***	-0.158***	0.266***	
	(0.008)	(0.019)	(0.019)	(0.036)	(0.049)	(0.049)	(0.005)	(0.018)	(0.018)	(0.015)	(0.031)	(0.031)	(0.009)	(0.017)	(0.017)	(0.011)	(0.022)	(0.023)	
Husband Active occupied, agriculture	0.097***	-0.098^^	0.001	0.282***	-0.138^^^	-0.144^^^	0.050***	0.026*	-0.076***	0.149***	-0.055^^	-0.094^^^	0.212***	-0.127***	-0.085^^^	0.191	-0.093***	-0.098^^^	
Husband Active accurried industry	(0.038)	(0.049)	(0.050)	(0.038)	(0.042)	(0.032)	(0.009)	(0.013)	(0.013)	(0.019)	(0.023)	(0.023)	(0.017)	(0.019)	(0.016)	(0.017)	(0.019)	(0.017)	
Husband Active occupied, industry	(0.020	(0.017)	(0.027	(0.027	(0.041)	(0.024	(0.002)	(0.003	-0.000	(0.022)	(0.002	(0.028)	(0.023	(0.028	(0.017)	(0.032	(0.020	(0.012)	
Husband Active occupied, trade	0.052***	0.028	-0.079***	0.022	-0.095**	0.072	0.003	0.028	-0.031*	-0.000	-0.016	0.017	-0.007	0.025	-0.018	0.056***	-0.018	-0.038**	
naobana nome eccapica, nade	(0.013)	(0.019)	(0.017)	(0.041)	(0.047)	(0.045)	(0.010)	(0.017)	(0.017)	(0.018)	(0.026)	(0.026)	(0.013)	(0.018)	(0.016)	(0.022)	(0.023)	(0.019)	
Husband Active occupied, public and social services	-0.043***	-0.021	0.064***	-0.027	-0.009	0.036	-0.007	0.033**	-0.026*	-0.030	-0.113***	0.142***	-0.026**	-0.004	0.030**	0.046**	0.014	-0.060***	
1 11	(0.012)	(0.020)	(0.019)	(0.045)	(0.048)	(0.041)	(0.009)	(0.015)	(0.015)	(0.028)	(0.036)	(0.038)	(0.011)	(0.016)	(0.015)	(0.023)	(0.023)	(0.018)	
Husband Active occupied, other services	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	Ref.	
Husband Active unemployed	0.271***	-0.169***	-0.103***	0.519***	-0.274***	-0.245***	0.416***	-0.156***	-0.261***	0.244***	-0.187***	-0.057	0.279***	-0.144***	-0.135***	0.329***	-0.201***	-0.128***	
	(0.037)	(0.035)	(0.033)	(0.109)	(0.103)	(0.027)	(0.026)	(0.025)	(0.017)	(0.037)	(0.037)	(0.036)	(0.036)	(0.033)	(0.024)	(0.042)	(0.036)	(0.022)	
Husband Inactive	0.077***	0.036*	-0.113***	0.450***	-0.285***	-0.165***	0.123***	-0.001	-0.121***	0.244***	-0.135***	-0.109***	0.047***	-0.131***	0.084***	0.145***	-0.107***	-0.037	
Luchand Na colformiousd	(0.016) Def	(0.019)	(0.017)	(0.078) Def	(0.062)	(0.039)	(0.013)	(0.018)	(0.020)	(0.040)	(0.038)	(0.031)	(0.018) Def	(0.022)	(0.022)	(0.036)	(0.035)	(0.029)	
Husband Selfemployed	Ker.	Ker.	Ker.	Ker. 0.160***	Ker.	Ker. 0.106***	Ker.												
nusbanu Sellemployeu	(0.011)	(0.005)	(0.013)	(0.023)	-0.053	(0.025)	(0.007	(0.013)	(0.014)	(0.013)	(0.012)	(0.017)	(0.040	(0.014)	(0.020	(0.010)	(0.033)	(0.013)	
	(0.011)	(0.010)	(0.010)	(0.020)	(0.020)	(0.020)	(0.000)	(0.010)	(0.014)	(0.010)	(0.017)	(0.017)	(0.011)	(0.014)	(0.014)	(0.010)	(0.010)	(0.010)	
Observations	22,549	22,549	22,549	2,548	2,548	2,548	41,738	41,738	41,738	8,064	8,064	8,064	18,271	18,271	18,271	14,063	14,063	14,063	

Standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1

Annex 6: Probit multinomial of variable poor/middle sectors/affluent for spouses